

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Smyth, Emer

Research Report The changing social worlds of 13-year-olds

Research Series, No. 178

Provided in Cooperation with: The Economic and Social Research Institute (ESRI), Dublin

Suggested Citation: Smyth, Emer (2024) : The changing social worlds of 13-year-olds, Research Series, No. 178, The Economic and Social Research Institute (ESRI), Dublin, https://doi.org/10.26504/rs178

This Version is available at: https://hdl.handle.net/10419/299345

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



WWW.ECONSTOR.EU

https://creativecommons.org/licenses/by/4.0/

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



RESEARCH SERIES NUMBER 178 MAY 2024

THE CHANGING SOCIAL WORLDS OF 13-YEAR-OLDS

EMER SMYTH





THE CHANGING SOCIAL WORLDS OF 13-YEAR-OLDS

Emer Smyth

MAY 2024

RESEARCH SERIES

NUMBER 178

Available to download from www.esri.ie

© The Economic and Social Research Institute Whitaker Square, Sir John Rogerson's Quay, Dublin 2

https://doi.org/10.26504/rs178



This Open Access work is licensed under a Creative Commons Attribution 4.0 International License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly credited.

ABOUT THE ESRI

The Economic and Social Research Institute (ESRI) advances evidence-based policymaking that supports economic sustainability and social progress in Ireland. ESRI researchers apply the highest standards of academic excellence to challenges facing policymakers, focusing on ten areas of critical importance to 21st Century Ireland.

The Institute was founded in 1960 by a group of senior civil servants led by Dr T.K. Whitaker, who identified the need for independent and in-depth research analysis. Since then, the Institute has remained committed to independent research and its work is free of any expressed ideology or political position. The Institute publishes all research reaching the appropriate academic standard, irrespective of its findings or who funds the research.

The ESRI is a company limited by guarantee, answerable to its members and governed by a Council, comprising up to 14 representatives drawn from a cross-section of ESRI members from academia, civil services, state agencies, businesses and civil society. Funding for the ESRI comes from research programmes supported by government departments and agencies, public bodies, competitive research programmes, membership fees, and an annual grant-in-aid from the Department of Public Expenditure NDP Delivery and Reform.

Further information is available at www.esri.ie.

THE AUTHOR

Emer Smyth is a Research Professor at the Economic and Social Research Institute (ESRI) and an Adjunct Professor at Trinity College Dublin (TCD).

ACKNOWLEDGEMENTS

This publication was funded by the Department of Children, Equality, Disability, Integration and Youth (DCEDIY) through a joint Research Partnership with the ESRI. The views, opinions, findings, conclusions and/or recommendations expressed here are strictly those of the author. They do not necessarily reflect the views of the DCEDIY, which takes no responsibility for any errors or omissions in, or for the accuracy of, the information contained in this publication. It is presented to inform and stimulate wider debate among the policy community and among academics and practitioners in the field. I am grateful to Grainne Collins, Clare Farrell, Ciara Pidgeon, Tony Fahey, Anna Visser and Aisling Murray for their advice and support for the project. Useful comments were provided by the external reviewer, two ESRI reviewers and the editor, Anne Nolan. As always, I am grateful to the *Growing Up in Ireland* (GUI) Study Team and the families involved in GUI for making this research possible.

Growing Up in Ireland (GUI) is the national longitudinal study of children, funded by Government. The project is managed and delivered through a collaboration between the DCEDIY and the Central Statistics Office (CSO). Results in this report are based on analyses of data from Research Microdata Files provided by the CSO and from Anonymised Microdata Files provided by the Irish Social Science Data Archive (ISSDA). Neither the CSO nor the DCEDIY takes any responsibility for the views expressed or the outputs generated from these analyses.

This report has been accepted for publication by the Institute, which does not itself take institutional policy positions. All ESRI Research Series reports are peer-reviewed prior to publication. The author(s) are solely responsible for the content and the views expressed.

TABLE OF CONTENTS

EXECUTIVE	SUMMA	ARY	X
Aim	of the s	study	x
Mai	in findin	gs	x
Imp	lications	s for policy	xi
CHAPTER 1:	INTROE	DUCTION	1
1.1	Backgr	ound to the study	1
1.2	The cha	anging context for adolescents	2
	1.2.1	The developmental context	2
	1.2.2	Policy and societal changes	3
	1.2.3	The changing profile of 13-year-olds and their families	4
1.3	Data ar	nd methodology	5
	1.3.1	Data collection	5
	1.3.2	Measures and methodology	6
1.4	Outline	e of the report	12
CHAPTER 2:	RELATIO	ONSHIPS WITH FAMILY AND PEERS	13
2.1	Introdu	uction	13
2.2	Relatio	onships with parents	13
2.3	Peer re	elationships	25
2.4	Summa	ary	31
CHAPTER 3:	DAY-TC	D-DAY ACTIVITIES	34
3.1	Introdu	uction	34
3.2	Structu	ured activities and physical exercise	34
3.3	Screen	time	46
3.4	Conclu	isions	55
CHAPTER 4:	EDUCA	TIONAL EXPERIENCES	58
4.1	Introdu	uction	58
4.2	School	engagement	58
4.3	Subject	t engagement	62
	4.3.1	Maths engagement	62
	4.3.2	Engagement in English	65
	4.3.3	Engagement in Irish	68
	4.3.4	Science engagement	71

4.4	Conclusions	74
CHAPTER 5:	CONCLUSIONS	. 79
5.1	Background to the study	79
5.2	Changes in social relationships	. 79
5.3	Day-to-day activities	82
5.4	Educational experiences	84
5.5	Implications for policy	84
REFERENCE	S	87

LIST OF TABLES

Table 1.1	Measures of parental and peer relationships7
Table 1.2	Measures of day-to-day activities9
Table 1.3	Measures of educational experiences11
Table 2.1	OLS regression models of maternal conflict with the 13-year-old (Pianta scale)14
Table 2.2	OLS regression models of paternal conflict with the 13-year-old (Pianta scale)17
Table 2.3	Logistic regression models of the factors associated with whether the young person
	gets on very well with their mother (odds ratios)19
Table 2.4	OLS regression models of the young person's perception of maternal responsiveness
Table 2.5	Logistic regression models of the family eating dinner together every day (odds ratios)
Table 2.6	Generalised linear models (cumulative logit function) of number of friends (odds ratios)
Table 2.7	Generalised linear model (cumulative logit function) of number of close friends
	(odds ratios)
Table 2.8	OLS regression models of SDQ Peer Problems subscale
Table A2.1	Sensitivity analyses of parent-child relationships (parent-reported) to include household
	income quintile
Table A2.2	OLS regression models of relationship between socio-emotional wellbeing and parent-
	child conflict
Table A2.3	Sensitivity analyses of peer group to include household income quintile
Table 3.1	Logistic regression models of the 13-year-old taking part in organised sports at least
	weekly (odds ratios)
Table 3.2	Logistic regression models of the 13-year-old having low levels of involvement in
	hard exercise (odds ratios)
Table 3.3	Logistic regression models of the 13-year-old having low levels of involvement in
	light exercise (odds ratios)40
Table 3.4	Logistic regression models of the 13-year-old having at least weekly involvement in
	cultural activities (odds ratios)42
Table 3.5	Logistic regression models of the 13-year-old having at least weekly involvement in
	organised clubs or groups (odds ratios)43
Table 3.6	Generalised linear models (cumulative logit function) of time spent watching television
	(odds ratios)48
Table 3.7	Generalised linear models (cumulative logit function) of time spent playing
	video/computer games (odds ratios)50
Table 3.8	Generalised linear models (cumulative logit function) of other screen time (odds
	ratios)
Table 3.9	Relationship between screen time (TV and other) and other day-to-day
	activities

Table A3.1	Sensitivity analyses of sports and cultural activities to include household income quintile
	(odds ratios)
Table A3.2	Sensitivity analyses of screen time to include household income quintile (odds ratios)
Table 4.1	Generalised linear models (cumulative logit function) of positive attitudes to school
	(odds ratios)61
Table 4.2	Multinomial logit models of attitude to Maths (odds ratios) (Base category: OK)63
Table 4.2	(Continued)
Table 4.3	Multinomial logit models of attitude to English (odds ratios) (Base category: OK) 66
Table 4.3	(Continued)
Table 4.4	Multinomial logit models of attitude to Irish (odds ratios) (Base category: OK)69
Table 4.4	(Continued)
Table 4.5	Multinomial logit models of attitude to Science (odds ratios) (Base category: OK)72
Table 4.5	(Continued)73
Table A4.1	Sensitivity analyses of attitude to school to include household income quintile
	(odds ratios)75
Table A4.2	OLS regression models of relationship between socio-emotional wellbeing and attitudes
	to school75
Table A4.3	Sensitivity analyses of attitude to Maths to include household income quintile
	(odds ratios)76
Table A4.4	Sensitivity analyses of attitude to English to include household income quintile
	(odds ratios)76
Table A4.5	Sensitivity analyses of attitude to Irish to include household income quintile (odds ratios)
Table A4.6	Sensitivity analyses of attitude to Science to include household income quintile
	(odds ratios)78
Table 5.1	Summary of patterns of change in adolescent experiences and outcomes between
	cohorts
Table 5.1	(Continued)

LIST OF FIGURES

Figure 1.1	Changes in family background characteristics between Cohorts '98 and '084
Figure 2.1	Predicted levels of mother-child conflict by gender, parental education, social class and financial strain
Figure 2.2	How parents deal with young person's misbehaviour by cohort, as reported by young people
Figure 2.3	Number of friends and of close friends by cohort (actual percentage)25
Figure 2.4	Mean SDQ peer problems by number of friends and of close friends by cohort29
Figure 3.1	Predicted probability of weekly involvement in organised sport by gender, parental education, social class and financial strain
Figure 3.2	Actual percentages with low levels of hard exercise by social class
Figure 3.3	Level of involvement in organised activities (sports, cultural and clubs) by cohort44
Figure 3.4	Time spent reading for pleasure by social class and gender, Cohort '98 (actual percentages)44
Figure 3.5	Frequency of reading for pleasure by social class and gender, Cohort '08 (actual percentages)45
Figure 3.6	Time spent on different types of screen-based activities by cohort46
Figure 3.7	Predicted probability of watching two or fewer hours TV per day by parental education and cohort47
Figure 3.8	Predicted probability of spending one or more hours online (excluding TV and video/computer games) per day by parental education and cohort51
Figure 4.1	School engagement by cohort (actual percentages)59
Figure 4.2	Perceived interest in Maths, English, Irish and Science by cohort (actual percentages)

ABBREVIATIONS

CSO	Central Statistics Office
GUI	Growing Up in Ireland
OLS	Ordinary Least Squares
SEN	Special Educational Needs
SDQ	Strengths and Difficulties Questionnaire

EXECUTIVE SUMMARY

AIM OF THE STUDY

This study draws on data on Growing Up in Ireland (GUI) Cohorts '98 and '08 to document changes in the lives of adolescents over the period 2011/12 to 2021/22, building on an earlier study (Smyth, 2022) which compared their experiences at nine years of age. This decade was a period of considerable social and policy change, including reform of the junior cycle, growing digitalisation and the disruption of the pandemic to all aspects of young people's lives. Changes were also evident in the profile of young people and their families, with increasing cultural diversity, higher education levels among parents, lower levels of financial strain and increasing numbers with a disability among members of Cohort '08 than among their older cohort counterparts.

The study looks at changes in 13-year-olds' relationships with their parents and peers, in their day-to-day activities and in their experiences of school. The main research questions addressed by the study are:

- 1. How have the quality of relationships, experience of learning and activities engaged in by adolescents changed over the course of a decade?
- 2. To what extent do any such changes reflect differences in the family characteristics of the young people?
- 3. Are any such changes more evident for boys or girls or for young people from different social backgrounds? Is differentiation by gender and social background in adolescents' social worlds less evident for the younger cohort than previously?

The remainder of this executive summary outlines the main findings of the study and discusses the implications for policy development.

MAIN FINDINGS

The study findings point to very significant changes in the lives of 13-year-olds, even taking account of the shift in their profile. Both mothers and fathers report much lower levels of conflict than previously while young people see their mothers as more responsive to their needs. Parents are also less likely to use more punitive approaches (such as grounding) to deal with behaviour and more likely to explain what the young person has done wrong. In contrast, the quality of peer engagement, at least as reported by mothers, has worsened over time, with greater peer problems evident among Cohort '08 members. Furthermore, young people report having smaller friendship groups than previously. It is not clear whether

these patterns reflect longer-term trends or the short-term impact of the pandemic.

Significant changes are evident in adolescent lives. There has been an increase in weekly involvement in organised sports and very low levels of engagement in hard and light exercise are less evident than a decade previously. Levels of weekly involvement in cultural activities (such as drama and dance) have been stable, if not increasing, though involvement in organised groups (like youth clubs and Scouts/Guides) has declined somewhat, at least partly because of ongoing pandemic restrictions at the time of the 2021/22 survey. Overall, the proportion of 13-year-olds who are not regularly involved in any of these structured activities has declined somewhat from 25 to 21 per cent. Unfortunately, no fully comparable information is available on changes in reading for pleasure but there is tentative evidence of a growth in the number of boys in particular who rarely read. Not surprisingly, there has been a shift away from traditional media (such as TV watching but also playing video/computer games) towards other (non-school-related) screen time. High levels of screen time are generally associated with less involvement in sport and cultural activities.

Cohort '08 had experienced the full implementation of junior cycle reform as well as a change in approaches to teaching and learning at both primary and secondlevel. This has apparently led to improved interest in English, Maths and, especially, Science. However, this has not translated in the same way into improved attitudes to school in general. Indeed, there is a decline in the proportion of girls who say they like school very much. Preliminary analysis suggests this is at least partly related to increased emotional difficulties among girls. Further research is merited on whether this reflects the impact of the pandemic or other societal changes.

IMPLICATIONS FOR POLICY

There are many positive findings emerging from this study, particularly more democratic family relations, improved engagement in sport and physical exercise, and greater interest in some core subjects in the wake of junior cycle reform. The study findings do, however, highlight persistent social background and gender differences in the lives adolescents lead. Financial strain continues to be a source of friction between parents and teenagers, reinforcing the need to target adequate levels of income support towards families with children. Young people from more disadvantaged backgrounds are less likely to take part in the kinds of out-of-school activities (like hard exercise and cultural engagement) that boost their development, highlighting the need for subsidised activities in communities and supports for schools to provide access to a range of extracurricular options.

There are marked gender differences in young people's lives, with girls more likely to be involved in cultural activities and reading and boys more likely to be involved in sport and hard physical exercise. Previous research suggests that these gendered patterns emerge early and in- and out-of-school settings should seek to provide all young people with access to a range of activities from their early years onwards. The findings point to a significant shift in screen time, with boys in Cohort '98 spending longer on other (non-TV, non-gaming) screen time than girls but this pattern reversing among Cohort '08, a pattern that merits further investigation. Gendered attitudes to subjects are also evident, with girls more positive about language-based subjects and boys more positive about Maths and Science. The widening gender gap in interest in STEM is concerning, given the policy effort in encouraging girls into STEM, and highlights the importance of inclusive curriculum and pedagogy from an early age.

At the age of nine (Smyth, 2022), 13-year-olds with a disability have much poorer outcomes across the main domains studied here. Further research is planned on the increase over the decade in the proportion with a disability and the implications for their experiences and outcomes.

The study was not designed to look at the effects of the pandemic on adolescent outcomes and, given the timing of data collection, it is not possible to discern whether any impact is long-lasting or not. Nonetheless, the findings point to poorer peer relations and more emotional difficulties among this cohort, especially among girls. There is tentative evidence that this gendered pattern of emotional difficulties is related to greater relative conflict with parents and less positive views of school among girls. Further research is merited to identify the risk and protective factors to inform policy development in a context where school principals report much poorer wellbeing and school attendance among students (Smyth, 2023) in the post-pandemic period.

CHAPTER 1

Introduction

1.1 BACKGROUND TO THE STUDY

The past decade has seen rapid social change in the lives of young people and their families, with increasing diversity in Irish society, growing digitalisation of daily lives, policy reform from early years to further and higher education, and the disruption to learning and wellbeing caused by the global pandemic. Having two cohorts in the Growing Up in Ireland (GUI) study enables detailed analyses of the extent of change in the lives of adolescents over a decade. This research builds upon a similar analysis of changes over time in the lives of nine-year-old children (Smyth, 2022), which showed improved parent-child relations but concerning declines in engagement in sports and growing socio-economic inequality in sports participation and reading. Information from the 13-year wave of Cohort '08 offers the potential to look at whether these changes persist into adolescence and whether new trends are observed in the wake of the pandemic. The domains analysed comprise relationships with parents and friends; the kinds of activities in which adolescents engage (including sports, cultural activities and screen time); and engagement in school.

The main research questions addressed by the study are:

- How have the quality of relationships, experience of learning and activities engaged in by adolescents changed over the course of a decade (2011/12– 2021/22)?
- 2. To what extent do any such changes reflect differences in the family characteristics of the young people?
- 3. Are any such changes more evident for boys or girls or for young people from different social backgrounds? Is differentiation by gender and social background in adolescents' social worlds less evident for the younger cohort than previously?

The remainder of this chapter places the study in the context of policy and broader social change, outlines the main changes in the profile of the 13-year-olds and discusses the data, measures and methodology used.

1.2 THE CHANGING CONTEXT FOR ADOLESCENTS

1.2.1 The developmental context

Adolescence is a time of important biological and social changes in young people's lives, with pubertal development and a growing importance of peer relationships (Brown and Larson, 2009). Adolescents have increasing independence from their parents, spending less time with them and engaging in fewer shared activities (Larson et al., 2013; Dubas and Gerris, 2002; Zimmer-Gembeck and Collins, 2006). Adolescence coincides with changes in educational experiences too. The transition to second-level education can lead to a disruption in existing peer networks, especially in Ireland where there is active school choice, and difficulties adjusting to the new school context with more subjects, several teachers and often larger schools (Smyth et al., 2004; Jindal-Snape et al., 2020, 2021). While there are common aspects to adolescent development, research has highlighted the way in which these developmental processes are shaped by broader social change (Crockett and Silbereisen, 2000; Schoon, 2006) and influenced by economic shocks (Conger et al., 2000).

Researchers internationally have pointed to the absence of systematic data on trends in the time use of adolescents (Hagell et al., 2012; Gracia, 2023). A notable exception is a body of work by Mullan (2019) who documented changes in the lives of 8–16-year-olds in the UK between 1974/75 and 2014/15. He found an increase between 1975 and 2000 in the time spent on sport by 11–13-year-olds but levels remained stable between 2000 and 2015. There was a decline in time spent on hobbies and outdoor play but the time spent on socialising outside the home was stable. Two main themes have emerged from other research. Firstly, studies have documented the increasing time spent on digital media and a decrease in time spent on traditional media such as television and books (see Twenge et al., 2019 on trends in the US 2010-2016; Patalay and Gage, 2019 on trends in England 2005-2015). Secondly, available research suggests an increase in emotional difficulties and mental health problems among teenagers over time (Hagell et al., 2012). In Ireland, the My World survey documented an increase in adolescent levels of depression and anxiety between 2010/11 and 2018/19, with a widening gender gap (Dooley et al., 2019). Several researchers have pointed to the interconnectedness of these two trends, with high levels of digital engagement seen to be associated with poorer wellbeing and greater mental health problems, especially for females (Twenge and Martin, 2020). However, other studies have contested this argument (see, for example, Orben and Przybylski, 2019) and instead have pointed to the role of increasing school-related stress over time in driving poorer wellbeing (Högberg et al., 2020; De Looze et al., 2020).

1.2.2 Policy and societal changes

Perhaps the most significant recent policy change affecting young people's lives in Ireland has been the reform of the junior cycle. The Framework for Junior Cycle (Department of Education and Skills, 2015) set out the plans for the phased implementation of reform, with changes in the content and methodology of subjects, a reduction in the number of subjects, the introduction of short courses, new programmes for students with special educational needs (SEN) and a new system of assessment and reporting. The framework had also specified a new curriculum area of wellbeing, with revised guidelines issued to schools in 2021 to emphasise the importance of a whole-school programme and underpinning climate to support wellbeing (NCCA, 2021). Cohort '08 members entered second-level education at a stage when the phasing of junior cycle implementation had been completed. An evaluation of the reform (McGarr et al., 2023) has indicated that students now experience a range of active approaches to teaching and learning but more teacher-directed approaches and an emphasis on individual work are still common. Despite the reduction in the number of subjects and greater emphasis on class-based assessments, students continue to report stress in balancing their workload and see exams as putting significant strain on their wellbeing (McGarr et al., 2023).

Members of Cohort '98 and of Cohort '08 were both exposed to massive societal shocks which differed in their timing and impact. Cohort '98 experienced the Great Recession in their middle childhood, with Ireland in economic recovery at the time they were surveyed at 13 years of age. Cohort '08 members were in infancy at the start of the recession. Research has shown that the recession resulted in increased levels of financial strain among families and poorer physical and socio-emotional wellbeing among children (Gibbons et al., 2023; Reinhard et al., 2018; Sprong et al., 2023). The COVID-19 pandemic and related public health restrictions led to massive disruption in the lives of young people from Cohort '08¹ and their families in terms of their learning, social relationships and day-to-day activities. The majority of 15-year-olds in Ireland reported a lack of motivation to learn and missing schoolorganised sports during the period of school closures, with higher levels doing so in Ireland than in the OECD as a whole (Donohue et al., 2023). Young people differed in their access to the resources to support home learning (GUI Study Team, 2021), with consequences for their wellbeing (Smyth and Murray, 2022). Two-thirds of Cohort '08 made the transition to second-level education between the two periods of school closures (GUI Study Team, 2021), with less preparation for the adjustment process than would have been the case previously. Less information has been available on how young people have fared since schools have reopened. In 2022, second-level school principals reported that a significant proportion of students were behind in their learning and, in most cases, wellbeing and attendance were worse than before the pandemic (Smyth, 2023). There has been a good deal of

¹ Cohort '98 experienced the pandemic at the time of key life transitions such as leaving school and entering further/higher education or the labour market. However, analyses in this report do not look at this phase.

research internationally looking at the effects of the pandemic on mental health and wellbeing during the period of disruption (see, for example, Newlove-Delgado et al., 2021; Hu and Qian, 2021). However, research is only emerging on whether these effects have persisted into the post-pandemic period.

In addition, there have been a number of other policy changes that would have impacted on the two cohorts differently, including the expansion of early years provision, changes in income support for families and the extension of entitlement to leave for parents (see Smyth and Russell, 2021).

1.2.3 The changing profile of 13-year-olds and their families

This section uses GUI data to outline the main changes in the profile of teenagers and their families over time. Reflecting large-scale educational expansion in Ireland, there is a marked increase in the proportion of 13-year-olds who have one or more parents with a degree, from 30 to 38 per cent (Figure 1.1). When Cohort '98 was 13 years of age, in 2011/12, Ireland was still in recovery from the Great Recession, with almost a quarter of families reporting difficulty or great difficulty making ends meet. By 2021/22, despite the temporary disruption to parental employment caused by the pandemic, the proportion was much lower at 10 per cent of families. In keeping with broader trends in housing tenure in Ireland, the proportion of 13-year-olds living in rented accommodation (either social housing or private rented) increased from 20 to 25 per cent. There has been significant growth in the level and nature of inward migration to Ireland in recent years (McGinnity et al., 2020), a pattern reflected in the increase in migrant-origin² young people from 8 per cent among Cohort '98 to 11 per cent among Cohort '08.



FIGURE 1.1 CHANGES IN FAMILY BACKGROUND CHARACTERISTICS BETWEEN COHORTS '98 AND '08

Source: GUI Cohorts '98 and '08.

² Migrant-origin young people comprise those for whom both of their parents were born outside Ireland. In the case of lone-parent families, classification is based on the birthplace of the resident parent.

There were fewer changes between cohorts in the composition of the family. The proportion of lone-parent families when the young person was 13 was broadly stable, declining only slightly from 19 to 18 per cent. Family size declined between cohorts, in keeping with national trends of a decline in the number of children per family (CSO, 2023);³ among Cohort '98, 22 per cent had three or more siblings but this declined to 15 per cent for Cohort '08. There was no shift in the distribution between urban and rural areas between cohorts.

The most significant difference between cohorts relates to the reported prevalence of having a long-term illness or disability. Mothers of Cohort '98 were asked two sets of questions about illness/disability: one was an open-ended question about whether the young person had a long-standing condition or disability while the other set of questions presented respondents with a list of conditions (such as physical disability or visual or hearing impairment, or specific learning disability) and asked whether the young person had any of those named (or another not specified). The incidence was higher using the list of conditions rather than the open-ended conditions. Mothers of Cohort '08 were not asked the open-ended question but were instead presented with a list of conditions (with a different classification to that used in Cohort '98). Comparing the 'list' reports, the prevalence increased from 19 per cent in Cohort '98 to 36 per cent in Cohort '08. Furthermore, there was an increase from 6 to 23 per cent in the proportion who indicated that the young person was 'severely' or 'to some extent' hampered in their daily activities by this condition. Future research will seek to unpack the comparability of measures of illness/disability over time and the factors underlying this marked increase in reported prevalence.

The significant changes in the socio-economic background of families, in family size and the prevalence of disability are taken into account in exploring the extent to which 13-year-olds have experienced changes in key domains of their lives (Section 1.3.2).

1.3 DATA AND METHODOLOGY

1.3.1 Data collection

Survey waves for Cohort '98 at 9, 13, 17/18 and 20 years of age and survey waves for Cohort '08 at 9 months, 3, 5 and 9 years of age involved face-to-face interviews with families, with interviews conducted with children and young people themselves from the age of nine onwards. More sensitive questions were administered by interviewers on a self-completion basis during the interview visit to the home. Fieldwork took place over the period August 2011 to February 2012. The period of fieldwork initiation for Cohort '08 at 13 years of age (July 2021 to June

³ https://www.cso.ie/en/releasesandpublications/ep/p-cpsr/censusofpopulation2022summaryresults/householdsizeandmaritalstatus/#:~:text=Census%202022%20recorded%201%2C841%2C152%20priv ate,average%2C%20which%20stood%20at%202.75.

2022) was one of continuing public health restrictions, a situation which necessitated a change in the mode used. The decision was taken to shift to a telephone survey with the self-complete element administered via an online platform hosted by the Central Statistics Office (Murray et al., 2023).

The change in mode had a number of implications for the survey content. Respondent burden would be greater over the phone so a number of questions initially planned for inclusion had to be dropped. More complex questions which required prompt cards were not feasible over the phone so could not be included. For several questions, simpler answer categories were required. The consequences of these changes for comparability between the two cohorts are discussed in Section 1.3.2. In addition, a time-use diary was not included for the young person. This meant that information could not be triangulated across research instruments where there was a change in question wording or answer categories (as was the case for reading for pleasure, for example).

For Cohort '08, a total of 10,052 13-year-olds and their families were targeted, made up of those who had taken part when the young person was nine years of age and a small proportion of those who had taken part in earlier rounds but not at nine. The response rate was 78 per cent of the valid sample (Murray et al., 2023). The response rate for Cohort '98 was higher at 90 per cent of the valid sample (Thornton et al., 2016). This change over time is likely to reflect the broader decline in response rates found in population surveys internationally (see, for example, Luiten et al., 2020) as well as the specific circumstances of the pandemic. In both survey waves, attrition was greater among more socio-economically disadvantaged groups so weighting is used to make the samples representative of the population as a whole.

1.3.2 Measures and methodology

The measures used in this study were selected to capture the key domains of young people's lives, including their relationships with parents and peers, their day-to-day activities and their engagement with school. Selection was also based on the comparability of question wording and the potential to recode answer categories to be consistent across the two cohorts.

	Cohort '98	Cohort '08	
Family relationships			
Mother-child conflict (parent report)	Conflict subscale of the short form of the Pianta Child-Parent Relationship Scale (8 items)	Conflict subscale of the short form of the Pianta Child-Parent Relationship Scale (8 items)	
Father-child conflict (parent report)	Conflict subscale of the short form of the Pianta Child-Parent Relationship Scale (8 items)	Conflict subscale of the short form of the Pianta Child-Parent Relationship Scale (8 items)	
Quality of mother-child relationship (young person report)	How well do you get on with your Mum? Very well/fairly well/ you and your Mum do not get on	We would now like to ask you some questions about the parent or guardian whom you live with and usually looks after you. Even if you live with two parents/guardians, please just keep one of them in mind when answering the next set of questions. How well do you get on with this parent or guardian who usually looks after you? Very well/fairly well/we do not get on	
Maternal responsiveness (young person report)	Subscale of Parenting Style Inventory II	Subscale of Parenting Style Inventory	
Dining together daily (parent report)	How many days per week do you sit down to eat together?	How many days per week do you sit down to eat together?	
Peer relationships			
Number of friends (young person report)	How many friends do you normally hang around with? <i>None; 1 or 2; 3–5; 6–10; more than 10</i>	How many friends do you normally hang around with? <i>None; 1 or 2; 3–5; 6–10; more than 10</i>	
Number of close friends (young person report)	How many of these would you describe as close friends?	How many of these would you describe as close friends?	
Peer problems (parent report)	Peer problems subscale of Strengths and Difficulties Questionnaire (SDQ)	Peer problems subscale of Strengths and Difficulties Questionnaire (SDQ)	

TABLE 1.1 MEASURES OF PARENTAL AND PEER RELATIONSHIPS

Source: GUI Cohorts '98 and '08.

Table 1.1 shows the measures used to capture relationships with parents and peers among 13-year-olds that are used in Chapter 2. Parents of Cohort '08 were asked about the level of conflict with the young person, using the Pianta Child-Parent Relationship scale, but not about levels of closeness. This reflected space constraints in the 13-year questionnaire for Cohort '08 and the lack of variability in the high levels of closeness found in earlier waves. For similar reasons, the measurement of the young person's perspective was limited to one subscale of the Parenting Style Inventory, responsiveness, and in relation to the primary caregiver only.⁴ Cohort '98 had also been administered the demandingness and autonomygranting subscales. Cohort '98 were asked how well they got on with their mothers and fathers. Cohort '08 were asked about how well they got on with the parent or

⁴ These were administered through the online, self-complete module so the total numbers responding were lower than for the phone interviews (see Murray et al., 2023).

guardian they live with and usually looks after them. Given the vast majority of primary caregivers in the study were female, these responses are taken to refer to the relationship with their mother.⁵ Questions on size of the peer network, overall and in terms of close friends, were comparable over time. Unfortunately, the questions on frequency of socialising with friends were not fully comparable and Cohort '08 members were not asked about their closeness with, and alienation from, their peers because these scales required a large number of items. Instead, the mother-reported peer problems subscale of the SDQ is used as a proxy for the quality of the young person's engagement with their peers. The wording of the subscale items means that they are likely to capture both how isolated the young person is from their peers (size of friendship network) and the quality of their engagement with them.

Table 1.2 shows the measures of day-to-day activities that are used in Chapter 3. Three measures of engagement in sports and physical activity are used. The question wording on involvement in organised sports (that is, those that involve a coach or instructor) is similar across cohorts but the response categories change. As a result, the analyses can distinguish only between those who take part weekly and all others. This lacks the fine-grained differentiation possible with the nineyear-old data where Smyth (2022) showed a decline over time in those involved in organised sports on an almost daily basis. In order to supplement this information, questions on involvement in hard and light exercise were used. The timeframe specified changed between cohorts, with a reference period of 14 days for Cohort '98 and 7 days for Cohort '08. Similarly, the intensity of engagement changed referring to 60 minutes of activity for Cohort '08 and 20 minutes of activity for Cohort '98. In addition, the answer categories are top-coded for Cohort '98 (referring to 9 or more days out of 14). For this reason, analyses focus on identifying those with very low levels of involvement (the equivalent of two days or fewer per fortnight) who are likely to be more comparable.

⁵ In addition, this measure is more strongly related to the quality of the relationship with mothers at the age of nine than the quality of the relationship with fathers.

Physical activity Organised sport (young person report) How often do you: play sports with a coach or instructor, or as part of an organised team, other than in PE class (swimming, soccer, hockey, etc.)? Levels of engagement in hard exercise How many times in the past 14 days oreather fast and make your heart beat faster? Over the past 7 days, on how many days were you physically active for a total of at least an hour (60 minutes) per day? (Increases your heave you done at least 20 minutes of exercise hard enough to make you breathe fast and make your heart beat faster? Levels of engagement in light exercise How many times in the past 14 days have you done at least 20 minutes of ight exercise hard enough to make you heart beat faster? None; 1-2; 3-5; 6-8; 9 or more days Over the past 7 days, on how many days did you take part in light exercise for a total of at least an hour (60 minutes) per day? (Not hard enough to make you breathe have you done at least 20 minutes of light exercise that was not hard enough to make you breathe fast and make your heart beat faster? None; 1-2; 3-5; 6-8; 9 or more days Over the past 7 days, on how many days did you take part in light exercise for a total of at least an hour (60 minutes) per day? (Not hard enough to make you breathe heavily or make your heart beat faster.) None; 2-1; 3-2; 3-4; 5; 6; 7 days/ every day Organised activities How often do you: take part in dance. dub, community or church groups? Never/less than once a week/1-3 times a week/4 or more times a week /3 creent time groups such as Guides or Scouts, youth club, community or church groups? Never/less than once a week/1-3 times a week/4 or more times a week /3 creent time		Cohort '98	Cohort '08
Organised sport (young person report)How often do you: play sports with a coach or instructor, or as part of an organised team, other than in PE class (swimming, soccer, hockey, ett.)? Never/less than once a week/1-a times a week/4 or more times a weekHow often do you: play sports with a coach or instructor, or as part of an organised activitiesLevels of engagement in hard exerciseHow many times in the past 14 days have you done at least 20 minutes of hard exerciseOver the past 7 days, on how many days were you physically active for a total of at least an hour (60 minutes) per day? (Increases your have you done at least 20 minutes of have you heart beat faster? None; 1-2; 3-5; 6-8; 9 or more daysCrean time tight exerciseHow often do you: take part in dame groups wich as Guides or Scouts, youth club, community or church groups? Never/less than once a week/at least on a more// structured duic, community or church groups? Never/less tha	Physical activity		
Levels of engagement in hard exerciseHow many times in the past 14 days have you done at least 20 minutes of you physically active for a total of at least an you physically active for a total of at least an hard exercise hard exercise hard enough to make you heart beat faster? None; 1-2; 3-5; 6-8; 9 or more daysOver the past 7 days, on how many days were 	Organised sport (young person report)	How often do you: play sports with a coach or instructor, or as part of an organised team, other than in PE class (swimming, soccer, hockey, etc.)? Never/less than once a week/1–3 times a week/4 or more times a week	How often do you: play sports with a coach or instructor, or as part of an organised team, other than in PE class (swimming, soccer, GAA games, hockey, etc.)? At least once a week/at least once a month/ less often or never
Levels of engagement in light exerciseHow many times in the past 14 days have you done at least 20 minutes of light exercise that was not hard enough to make you breathe fast and make your heart beat faster? None; 1–2; 3–5; 6–8; 9 or more daysOver the past 7 days, on how many days did you take part in light exercise for a total of at least an hour (60 minutes) per day?Organised activitiesMow often do you: take part in dance, drama or music lessons? Never/less than once a week/1–3 times a week/4 or more times a weekHow often do you: Take part in art, crafts, drama or music lessons? At least once a week/at least once a month/ less often or neverInvolvement in organised groupsHow often do you: take part in clubs or groups such as Guides or Scouts, youth club, community or church groups? Never/less than once a week/1–3 times 	Levels of engagement in hard exercise	How many times in the past 14 days have you done at least 20 minutes of exercise hard enough to make you breathe fast and make your heart beat faster? None; 1–2; 3–5; 6–8; 9 or more days	Over the past 7 days, on how many days were you physically active for a total of at least an hour (60 minutes) per day? (Increases your heart rate and makes you get out of breath some of the time.) <i>None/zero days; 1; 2; 3; 4; 5; 6; 7 days/</i> <i>every day</i>
Organised activitiesHow often do you: take part in dance, drama or music lessons?How often do you: Take part in art, crafts, drama or music lessons?activitiesHow often do you: take part in dance, drama or music lessons?How often do you: Take part in art, crafts, drama or music lessons?activitiesNever/less than once a week/1–3 times a week/4 or more times a weekHow often do you: take part in clubs or groups such as Guides or Scouts, youth club, community or church groups? Never/less than once a week/1–3 times a week/4 or more times a weekHow often do you: take part in clubs or groups such as Guides or Scouts, youth club, community or church groups? Never/less than once a week/1–3 times a week/4 or more times a weekHow often do you: take part in clubs or groups such as Guides or Scouts, youth club, community or church groups? Never/less than once a week/1–3 times a week/4 or more times a weekOn a week/at least once a month/ less often or neverScreen timeOn a normal weekday during termtime, about how many hours do you spend watching television, videos or DVDs? Hours/minutesOn a weekday during termtime, how much time do you spend: watching television/ films/videos (on TV set, tablet or other device) None; <1; 1–2; 2–3; 3–4; 5 hours or moreVideo/computer gamingOn a normal weekday during termtime, about how many hours do you spend 	Levels of engagement in light exercise	How many times in the past 14 days have you done at least 20 minutes of light exercise that was not hard enough to make you breathe fast and make your heart beat faster? None; 1–2; 3–5; 6–8; 9 or more days	Over the past 7 days, on how many days did you take part in light exercise for a total of at least an hour (60 minutes) per day? (Not hard enough to make you breathe heavily or make your heart beat faster.) <i>None/zero days; 1; 2; 3; 4; 5; 6; 7 days/</i> <i>every day</i>
Structured cultural activitiesHow often do you: take part in dance, drama or music lessons? Never/less than once a week/1-3 times a week/4 or more times a weekHow often do you: Take part in art, crafts, drama or music lessons/clubs/rehearsals? Take part in dance lessons? At least once a week/at least once a month/ less often or neverInvolvement in organised groupsHow often do you: take part in clubs or groups such as Guides or Scouts, youth club, community or church groups? Never/less than once a week/1-3 times a week/4 or more times a weekHow often do you: take part in clubs or groups such as Guides or Scouts, youth club, community or church groups? Never/less than once a week/1-3 times a week/4 or more times a weekHow often do you: take part in clubs or groups such as Guides or Scouts, youth community or church groups? At least once a week/at least once a month/ less often or neverScreen timeOn a normal weekday during termtime, about how many hours do you spend watching television, videos or DVDs? Hours/minutesOn a weekday during termtime, how much time do you spend: watching television/ 	Organised activities		
Involvement in organised groupsHow often do you: take part in clubs or groups such as Guides or Scouts, youth 	Structured cultural activities	How often do you: take part in dance, drama or music lessons? Never/less than once a week/1–3 times a week/4 or more times a week	How often do you: Take part in art, crafts, drama or music lessons/clubs/rehearsals? Take part in dance lessons? At least once a week/at least once a month/ less often or never
Screen timeTelevision watchingOn a normal weekday during termtime, about how many hours do you spend watching television, videos or DVDs? Hours/minutesOn a weekday during termtime, how much time do you spend: watching television/ films/videos (on TV set, tablet or other device) None; <1; 1-2; 2-3; 3-4; 5 hours or moreVideo/computer gamingOn a normal weekday during termtime, about how many hours do you spend playing video games such as PlayStation, Xbox, Nintendo, etc.? Hours/minutesOn a weekday during termtime, ames None; <1; 1-2; 2-3; 3-4; 5 hours or moreOther screen timeOn a normal weekday during termtime, about how many hours do you spend using the computer? Hours/minutesOn a weekday during termtime, based activities? None; <1: 1-2: 2-3; 3-4: 5 hours or more	Involvement in organised groups	How often do you: take part in clubs or groups such as Guides or Scouts, youth club, community or church groups? Never/less than once a week/1–3 times a week/4 or more times a week	How often do you: take part in clubs or groups such as Guides or Scouts, youth club, community or church groups? At least once a week/at least once a month/ less often or never
Television watchingOn a normal weekday during termtime, about how many hours do you spend watching television, videos or DVDs? Hours/minutesOn a weekday during termtime, do you spend: watching television/ films/videos (on TV set, tablet or other device) None; <1; 1-2; 2-3; 3-4; 5 hours or more	Screen time		
Video/computer gamingOn a normal weekday during termtime, about how many hours do you spend playing video games such as PlayStation, Xbox, Nintendo, etc.? Hours/minutesOn a weekday during termtime, how much time do you spend: playing video/computer games None; <1; 1-2; 2-3; 3-4; 5 hours or more	Television watching	On a normal weekday during termtime, about how many hours do you spend watching television, videos or DVDs? <i>Hours/minutes</i>	On a weekday during termtime, how much time do you spend: watching television/ films/videos (on TV set, tablet or other device) <i>None;</i> <1; 1–2; 2–3; 3–4; 5 hours or more
Other screen timeOn a normal weekday during termtime, about how many hours do you spend using the computer?On a weekday during termtime, how much time do you spend on: other online or screen- based activities?Hours/minutesNone: <1: 1-2: 2-3: 3-4: 5 hours or more	Video/computer gaming	On a normal weekday during termtime, about how many hours do you spend playing video games such as PlayStation, Xbox, Nintendo, etc.? <i>Hours/minutes</i>	On a weekday during termtime, how much time do you spend: playing video/computer games None; <1; 1-2; 2-3; 3-4; 5 hours or more
	Other screen time	On a normal weekday during termtime, about how many hours do you spend using the computer? <i>Hours/minutes</i>	On a weekday during termtime, how much time do you spend on: other online or screen- based activities? None; <1; 1–2; 2–3; 3–4; 5 hours or more

TABLE 1.2 MEASURES OF DAY-TO-DAY ACTIVITIES

In terms of other structured activities, the question on involvement in organised groups remained the same, though the answer categories changed so, as with sport, the analyses can distinguish only between weekly involvement and all others. There was some change to the recording of structured cultural activities in Cohort '08, with the addition of arts/crafts and of clubs/rehearsals to the question wording. In addition, involvement in dance lessons was asked about separately. As with sports, the analyses can only distinguish between those who take part weekly and all others. The change in wording for cultural activities may result in a somewhat higher number for Cohort '08 but will give insights into the patterning by gender and social background, an important issue given inequalities evident at the age of nine.

A limitation in the information on cultural engagement is the lack of comparability of the questions on reading. Cohort '98 members were asked about how much time they read for pleasure per day, with the option of recording no time, while Cohort '08 members were asked about the frequency per week. Nonetheless, side-by-side comparisons and the patterning by gender and social background provide useful insights into reading behaviour among the two cohorts (see Chapter 3).

Previous research on the two cohorts at the age of nine had shown increasing levels of mobile phone ownership and a shift in the type of screen time (Smyth, 2022). For the two cohorts at age 13, we can distinguish between time spent watching TV/films/videos (on a TV set, tablet or other device), time spent playing video/computer games and other online or screen-based activities. Cohort '98 were asked to record the exact number of hours/minutes so these have been regrouped to be comparable with the responses among Cohort '08. Given the changes over time, it might be the case that there is some blurring of the boundaries between the categories; for example, some types of gaming might be counted by the young person as 'other' screen time rather than computer/video gaming. Nonetheless, the analyses can point to important differences between groups of young people in their digital engagement.

TABLE 1.3	MEASURES OF EDUCATIONAL I	EXPERIENCES
_		

	Cohort '98	Cohort '08
School engagement (young person report) Interest in Maths (young person report)	How do you feel about school in general? Like it very much/like it quite a bit/like it a bit/don't like it very much/hate it For each of these subjects, please indicate if you find the subject: interesting, OK, not interesting or you, don't take the subject.	How do you feel about school in general? Like it very much/like it quite a bit/like it a bit/ don't like it very much/hate it For each of these subjects, please indicate if you find the subject: interesting, OK, not interesting or you, don't take the subject.
Interest in English (young person report)	For each of these subjects, please indicate if you find the subject: <i>interesting</i> , OK, not interesting or you, don't take the subject.	For each of these subjects, please indicate if you find the subject: <i>interesting</i> , OK, not interesting or you, don't take the subject.
Interest in Irish (young person report)	For each of these subjects, please indicate if you find the subject: interesting, OK, not interesting or you, don't take the subject.	For each of these subjects, please indicate if you find the subject: interesting, OK, not interesting or you, don't take the subject.
Interest in Science (young person report)	For each of these subjects, please indicate if you find the subject: interesting, OK, not interesting or you, don't take the subject.	For each of these subjects, please indicate if you find the subject: interesting, OK, not interesting or you, don't take the subject.

Source: GUI Cohorts '98 and '08.

The measures of educational experiences discussed in Chapter 4 are outlined in Table 1.3. The measures were selected to capture overall school engagement and interest in key subject areas in the wake of junior cycle reform as well as the effect of the pandemic. The measures are fully comparable across cohorts.

The analyses explore differences between cohorts in their behaviours and experiences and also examine whether these differences between cohorts are apparent when we take account of changes over time in their composition. These analyses include a range of individual and family background factors that are measured in a fully comparable way between cohorts, namely, gender, parental education, household social class, whether the family is a lone-parent family, whether they have a migrant background (with both parents born abroad), whether the young person has three or more siblings, whether the 13-year-old has a longterm illness or disability (as reported by their mother), and whether they live in a rural or urban area. Although there are a number of measures of family background in the analyses, an additional measure is included on whether the family lives in rented accommodation (that is, in the private rented sector or renting from a local authority or approved housing body). This sector was previously found to have poorer quality housing on average and a negative impact on a range of child outcomes (see Laurence et al., 2023). In the appendix tables in Chapters 2 to 4, further analyses are included which examine the relationship with household income, measured in quintiles (fifths).

The analyses are based on the 13,581 cases that took part in all waves of the respective cohorts: 6,056 from Cohort '98 and 7,525 from Cohort '08. To retain as many cases as possible, those who do not report household income are assigned to a separate category and included in the analyses. The number of cases varies slightly across analyses due to differential missing information on the outcome variables. The number of cases for father-reported outcomes is smaller as non-resident fathers were not surveyed and not all resident fathers completed the secondary caregiver questionnaire (for further detail, see Smyth and Russell, 2021). Weights are used to account for non-response and attrition. The two datasets were pooled in order to directly test between-cohort differences.

The methodology used is similar to that employed to analyse change among nineyear-olds (Smyth, 2022). In the analyses presented in Chapters 2 to 4, a series of nested regression models are conducted. Model 1 compares the outcome for Cohort '98 and Cohort '08 to provide a measure of the raw difference between the two cohorts. Model 2 adds in the young person and family characteristics measured in both cohorts (outlined in the previous paragraph) to take account of potential changes in composition over time. In other words, these analyses allow us to determine, for example, whether improved sports participation over time is due to the higher educational levels found among parents of Cohort '08 young people. Model 3 includes interaction terms for gender and parental education; this allows us to assess whether gender and educational differences change between cohorts. Model 4 includes interaction terms for social class and financial strain to examine whether the social class gap and gap by economic insecurity have changed over time. Appendix tables in Chapters 2 to 4 present sensitivity analyses to explore whether there are changes in the impact of household income over time.

Because interaction terms can be difficult to interpret in an intuitive way, for selected outcomes, graphs are used to depict the interaction between cohort and, say, parental education. Thus, the graph shows the relationship between parental education and the outcome in question, showing how this relationship changes over time, holding the other factors constant.

1.4 OUTLINE OF THE REPORT

Chapter 2 looks at the factors associated with (changes in) family and peer relationships while Chapter 3 examines the extent to which day-to-day activities, including sports, cultural activities and screen time, have changed between cohorts. Chapter 4 looks at school engagement and level of interest in key school subjects. Chapter 5 outlines the main findings and discusses the implications for policy development.

CHAPTER 2

Relationships with family and peers

2.1 INTRODUCTION

This chapter looks at changes over time in the nature of relationships that adolescents have with their parents and peers. Multivariate models are used to examine whether between-cohort differences in relationships are evident when we take account of the changing profile of 13-year-olds over time. In addition, the models assess whether gender and social background differences have increased or decreased over time. Section 2.2 looks at the extent of parent-child conflict, reported by mothers and fathers respectively, and the responsiveness of mothers, as reported by the young person. The quality of the family relationship is also examined by looking at whether families eat dinner together on a daily basis. Section 2.3 looks at the size of the peer group, distinguishing between all friends and close friends; the Strengths and Difficulties Questionnaire (SDQ) subscale on peer problems, as reported by the mother, is used as a proxy for friendship quality.

2.2 RELATIONSHIPS WITH PARENTS

Tables 2.1 and 2.2 show the model coefficients for mother-child and father-child conflict respectively, using the Pianta measure of the degree of conflict, with higher scores indicating more conflict. The values for the scale vary from 8 to 40, with a (pooled) mean of 14.8 and a standard deviation of 6.7 for mother-child conflict (and 14.3 and 6 respectively for father-child conflict). There is a significant reduction in the level of mother-child conflict over time, with a raw difference of over one point (around a sixth of a standard deviation) between Cohort '98 and Cohort '08 (Model 1, Table 2.1). The between-cohort difference becomes larger when individual and family background factors are included in the analyses (Model 2); in other words, the decline in mother-child conflict is greater than might have been expected given the changes in young people's profile. There is no systematic variation in conflict by migrant status or household income (Table A2.1). Levels of conflict are found to be higher in lone-parent families, among those living in rented accommodation and those living in urban areas while they are lower in the case of larger families (where young people have three or more siblings). Experiencing financial strain (difficulty or great difficulty making ends meet) is associated with much higher levels of conflict. The strongest relationship is with having a disability, where levels of mother-child conflict are much higher than for other groups. To contextualise the scale of the change over time, it is worth noting that the between-cohort difference is on par with the size of the impact of financial strain and only exceeded in size by disability status.

TABLE 2.1OLS REGRESSION MODELS OF MATERNAL CONFLICT WITH THE 13-YEAR-OLD
(PIANTA SCALE)

	Raw differences (1)	Differences controlling for family and child factors (2)	Changes in the effect of gender and education (3)	Changes in the effect of class and strain (4)
Constant	15.293	13.740	14.214	13.920
Cohort '08	-1.121***	-1.574***	-2.693***	-1.920***
(Ref. Cohort '98)				
Female		0.505***	0.108	0.104
(Ref. Male)				
Parental education:				
Leaving Certificate		-0.251	-0.466±	-0.228
Post-secondary		0.057	-0.193	0.053
Degree		0.162	-0.244	0.124
(Ref. Lower secondary)				
Household social class:				
Professional		0.464±	0.439	0.331
Managerial		-0.001	-0.026	-0.041
Other non-manual		0.203	0.161	0.086
Skilled manual		-0.130	-0.166	-0.029
Non-employed		0.121	0.113	0.927**
(Ref. Semi/unskilled manual)				
Experiencing financial strain		1.468***	1.463***	1.292***
Parent(s) born outside Ireland		0.272	0.272	0.292
Lone-parent family		0.546***	0.535***	0.502**
(Ref. Two-parent family)				
Large family		-0.394**	-0.388**	-0.374**
(Ref. 2 or fewer siblings)				
Disability/illness of young person		3.063***	3.054***	3.051***
Urban location		0.504***	0.497***	0.490**
(Ref. Rural)				
Social/private rented tenure		0.395*	0.417*	0.431**
(Ref. Own with/without mortgage)				
Female* Cohort '08			0.859***	0.886***
Leaving Certificate* Cohort '08			0.570	
Post-secondary* Cohort '08			0.695±	
Degree* Cohort '08			1.004*	
Professional* Cohort '08				0.270
Managerial* Cohort '08				0.112
Non-manual* Cohort '08				0.222
Skilled* Cohort '08				-0.299
Non-employed* Cohort '08				-1.923***
Strain* Cohort '08				1.543*
Adjusted R ²	0.007	0.064	0.065	0.067
N (unweighted)		14,:	158	

 Source:
 Growing Up in Ireland Cohorts '98 and '08.

 Note:
 *** p<.001; ** p<.01; * p<.05; + p<.10.</td>

As well as looking at overall changes in levels of conflict, we are interested in the extent to which the relationship between individual and background factors and conflict shifts over time. Whether the gender or social gradient has changed can be assessed by looking at the significance level of the interaction term. For ease of interpretation, the predicted values are presented in Figure 2.1.





It is clear from Figure 2.1 that levels of maternal conflict declined over time for all of the specified groups. However, the extent of decline varied across gender and social groups (Figure 2.1). The decline in mother-son conflict was greater than for mothers and daughters, with the gender gap thus increasing over time. For Cohort '98, girls have only very slightly higher levels of conflict than boys but by Cohort '08, there is a much clearer gender difference (around a sixth of a standard deviation). Additional analyses explored what might account for this shift in the pattern by gender. The prevalence of emotional difficulties (measured by mother reports on the SDQ subscale)⁶ increased between Cohorts '98 and '08, with a greater increase for girls; this may reflect the impact of the pandemic and/or longer-term shifts in wellbeing.⁷ Table A2.2 presents a very simple model looking at the relationship with gender and cohort taking account of SDQ emotional difficulties. Not surprisingly, mother-child conflict is higher where young people have greater emotional difficulties, though it is not possible with these analyses to determine the direction

Source: Derived from Models 3 and 4 in Table 2.1.

⁶ The range for the SDQ emotional subscale is 0 to 10; the pooled mean is 2.1 with a standard deviation of 2.2.

⁷ Further research is needed to disentangle the potential drivers of this increase.

of causality. Nonetheless, the level of emotional difficulties is found to account for about two-thirds of the relative increase in mother-daughter conflict over time (compare Models 1 and 2). In other words, the fact that girls from Cohort '08 have more conflict with their mothers than boys do is at least partly related to their higher levels of emotional difficulties.

Conflict declined less for more advantaged families (professional or graduate parents) than for more disadvantaged groups (with parents with lower secondary (Junior Certificate) education or not in employment). However, the pattern for experience of financial strain is quite different. There was only a slight decline in conflict for those whose families had difficulties making ends meet while there was a much larger decline for other groups. Thus, financial strain has a stronger relationship with maternal conflict among Cohort '08.

OLS REGRESSION MODELS OF PATERNAL CONFLICT WITH THE 13-YEAR-OLD (PIANTA **TABLE 2.2** SCALE)

	Raw differences (1)	Differences controlling for family and child factors (2)	Changes in the effect of gender and education (3)	Changes in the effect of class and strain (4)
Constant	15.062	13.893	14.180	13.548
Cohort '08	-1.767***	-2.131***	-2.944***	-1.387**
(Ref. Cohort '98)				
Female		0.531***	0.286±	0.287±
(Ref. Male)				
Parental education:				
Leaving Certificate		-0.301	-0.499	-0.255
Post-secondary		-0.315	-0.315	-0.254
Degree		0.293	0.017	0.306
(Ref. Lower secondary)				
Social class:				
Professional		0.602*	0.586*	0.861*
Managerial		0.150	0.133	0.472
Other non-manual		0.236	0.228	0.743*
Skilled manual		-0.162	-0.176	0.439
Non-employed		0.435	0.428	2.038***
(Ref. Semi/unskilled manual)				
Experiencing financial strain		0.779***	0.778***	0.707***
Parent(s) born outside Ireland		-0.301	-0.306	-0.216
Large family		-0.108	-0.095	-0.081
(Ref. 2 or fewer siblings)				
Disability/illness		1.961***	1.961***	1.931***
Urban location		0.718***	0.712***	0.725***
(Ref. Rural)				
Social/private rented tenure		1.000***	1.009***	1.022***
(Ref. Own with/without mortgage)				
Female* Cohort '08			0.548*	0.554*
Leaving Certificate* Cohort '08			0.812	
Post-secondary* Cohort '08			0.264	
Degree* Cohort '08			0.781	
Professional* Cohort '08				-0.679
Managerial* Cohort '08				-0.770±
Non-manual* Cohort '08				-1.243*
Skilled* Cohort '08				-1.520**
Non-employed* Cohort '08				-3.851***
Strain* Cohort '08				0.286
Adjusted R ²	0.022	0.055	0.056	0.059
Ν		10,4	147	

 Source:
 Growing Up in Ireland Cohorts '98 and '08.

 Note:
 *** p<.001; ** p<.01; * p<.05; + p<.10.</td>

Table 2.2 shows the equivalent patterns for conflict between 13-year-olds and their fathers. This information is only available for resident fathers (see Chapter 1). There is a somewhat larger decline in father-child conflict than was the case for mothers, with levels dropping by over a quarter of a standard deviation over time (Model 1). Again, we see that the decline is greater than might be expected given the changes in profile of young people and their families (Model 2). Paternal conflict does not vary significantly by parental education, migrant status, family size (Table 2.2) or income (Table A2.1). As with maternal conflict, levels are higher for young people with a disability, those living in rented accommodation and those living in urban areas (Model 2). Young people whose families experience financial strain have higher levels of conflict with their fathers but this pattern does not change over time. As with maternal conflict, conflict declines more for boys than for girls, widening the gender gap (Model 3). Among the later cohort, fathers have higher levels of conflict with their daughters than with their sons, a pattern that appears to be accounted for by rising levels of emotional difficulties among girls (Appendix Table A2.1). Conflict declines over time across all social class groups, with the greatest decline among non-employed households (Model 4).

Although parent-child conflict had decreased over time, descriptive analyses show a slight decline in the proportion of young people who reported getting on very well with their mother, from 79 to 77 per cent. The factors associated with this outcome are presented in Table 2.3. Unlike mother-child conflict, the outcome is binary, distinguishing between those who report getting on very well with their mother and all others. A logistic regression model is therefore used. The coefficients are presented in terms of odds ratios; values above one indicate that a factor is associated with increased chances of getting on very well with their mother while values below one indicate reduced chances. Analyses indicate that the small decline found descriptively is accounted for by a shift in the profile of adolescents and their families, with no change over time when cohort composition is taken into account (compare the cohort coefficient in Models 1 and 2, Table 2.3). Girls are less likely to report very good relationships with their mothers and this gender gap widens significantly between cohorts. Young people whose parents have post-secondary or tertiary qualifications are less likely to report a very good relationship than those whose parents have lower levels of education; relationship quality is also somewhat lower among those from the highest-income families (Table A2.1). Variation by social background does not vary markedly over time, though relationships improve more for the non-employed group than for others. Relationships are somewhat less positive among those from lone-parent families, larger families and those with a disability.

TABLE 2.3LOGISTIC REGRESSION MODELS OF THE FACTORS ASSOCIATED WITH WHETHER
THE YOUNG PERSON GETS ON VERY WELL WITH THEIR MOTHER (ODDS RATIOS)

	Raw differences (1)	Differences controlling for family and child factors (2)	Changes in the effect of gender and education (3)	Changes in the effect of class and strain (4)	
Constant	3.745	1.907	1.880	6.501	
Cohort '08	0.878*	0.923	0.837	1.007	
(Ref. Cohort '98)					
Female		0.692***	0.796***	0.801***	
(Ref. Male)					
Parental education:					
Leaving Certificate		0.877	0.841±	0.871	
Post-secondary		0.791*	0.710**	0.802*	
Degree		0.661***	0.628***	0.671***	
(Ref. Lower secondary)					
Social class:					
Professional		1.047	1.060	0.985	
Managerial		1.014	1.023	0.970	
Other non-manual		0.960	0.969	0.970	
Skilled manual		1.063	1.073	1.000	
Non-employed		1.003	1.018	0.797±	
(Ref. Semi/unskilled manual)					
Experiencing financial strain		0.929	0.928	0.943	
Parent(s) born outside Ireland		0.925	0.944	0.937	
Lone-parent family		0.735***	0.745***	0.748***	
(Ref. Two-parent family)					
Large family		0.840**	0.837**	0.834**	
(Ref. 2 or fewer siblings)					
Disability/illness		0.832***	0.837**	0.838**	
Urban location		0.956	0.951	0.957	
(Ref. Rural)					
Social/private rented tenure		0.995	0.980	0.992	
(Ref. Own with/without mortgage)					
Female* Cohort '08			0.617***	0.611***	
Leaving Certificate* Cohort '08			1.293		
Post-secondary* Cohort '08			1.633*		
Degree* Cohort '08			1.422±		
Professional* Cohort '08				1.219	
Managerial* Cohort '08				1.143	
Non-manual* Cohort '08				0.989	
Skilled* Cohort '08				1.263	
Non-employed* Cohort '08				2.461***	
Strain* Cohort '08				0.919	
Nagelkerke R ²	0.001	0.020	0.024	0.026	
Ν	10,109				

 Source:
 Growing Up in Ireland Cohorts '98 and '08.

 Note:
 *** p<.001; ** p<.01; * p<.05; + p<.10.</td>

Table 2.4 looks at the young person's views on the responsiveness of their mother. The responsiveness scale varies from 5 to 25, with higher scores indicating a better relationship, with a (pooled) mean of 20.7 and a standard deviation of 3.4. Consistent with the decline in parent-child conflict, responsiveness levels improved significantly over time. This difference is over a fifth of a standard deviation and is larger than the effect of any individual or background factors. Young people from a migrant-origin family, those in larger families and those with a disability were significantly less likely to see their mother as responsive. In contrast to the parentreported patterns, there was no significant variation by urban/rural location, housing tenure or family structure. Perceived responsiveness increased over time for both girls and boys but there was a larger increase for boys. Among Cohort '98, girls were more likely than boys to perceive their mothers as responsive but this pattern had reversed by Cohort '08. Responsiveness increased for all social groups but the increases were somewhat larger among those whose parents had Junior Certificate education only or whose parents were not in employment. Among Cohort '08 members, there is little systematic variation by parental education or social class, except for lower levels of responsiveness among the semi/unskilled manual group than other social classes. Similarly, responsiveness improved somewhat more for those in the lowest income groups (Table A2.1).

TABLE 2.4 OLS REGRESSION MODELS OF THE YOUNG PERSON'S PERCEPTION OF MATERNAL RESPONSIVENESS

	Raw differences (1)	Differences controlling for family and child factors (2)	Changes in the effect of gender and education (3)	Changes in the effect of class and strain (4)	
Constant	20.484	20.589	20.261	20.523	
Cohort '08	0.754***	0.766***	1.919***	0.751***	
(Ref. Cohort '98)					
Female		0.344***	0.658***	0.655***	
(Ref. Male)					
Parental education:					
Leaving Certificate		-0.086	0.104	-0.050	
Post-secondary		0.025	0.054	0.053	
Degree		-0.214	-0.022	-0.166	
(Ref. Lower secondary)					
Social class:					
Professional		0.246	0.272	0.096	
Managerial		0.234±	0.243±	0.109	
Other non-manual		0.089	0.108	-0.067	
Skilled manual		0.049	0.060	-0.192	
Non-employed		0.230	0.245	-0.028	
(Ref. Semi/unskilled manual)					
Experiencing financial strain		-0.130	-0.117	-0.119	
Parent(s) born outside Ireland		-0.469***	-0.452***	-0.442***	
Lone-parent family		0.017	0.048	0.059	
(Ref. Two-parent family)					
Large family		-0.423***	-0.416***	-0.427***	
(Ref. 2 or fewer siblings)					
Disability/illness		-0.421***	-0.418***	-0.397***	
Urban location		-0.085	-0.075	-0.084	
(Ref. Rural)					
Social/private rented tenure		-0.184±	-0.198±	-0.209*	
(Ref. Own with/without mortgage)					
Female* Cohort '08			-1.147***	-1.185***	
Leaving Certificate* Cohort '08			-1.168***		
Post-secondary* Cohort '08			-0.335		
Degree* Cohort '08			-0.772*		
Professional*Cohort '08				0.592±	
Managerial* Cohort '08				0.464±	
Non-manual* Cohort '08				0.645*	
Skilled* Cohort '08				0.991*	
Non-employed* Cohort '08				1.026**	
Strain* Cohort '08				-0.071	
Adjusted R ²	0.009	0.020	0.027	0.026	
Ν	9,881				

 Source:
 Growing Up in Ireland Cohorts '98 and '08.

 Note:
 *** p<.001; ** p<.01; * p<.05; + p<.10.</td>



FIGURE 2.2 HOW PARENTS DEAL WITH YOUNG PERSON'S MISBEHAVIOUR BY COHORT, AS REPORTED BY YOUNG PEOPLE

Source: GUI Cohort '98 and Cohort '08 at 13 years of age.

Note: C98: Cohort '98; C08: Cohort '08.

Consistent with this reduction in parent-child conflict and improvement in maternal responsiveness, the ways in which parents deal with misbehaviour have also changed. The analyses presented here are descriptive to indicate the types of parent-child interaction which may underlie the findings on conflict and responsiveness discussed above. Figure 2.2 shows a marked increase in the regular use of explaining to the young person what they have done wrong (from 49% to 63%). There is also a marked reduction in parents shouting at the young person and in the use of grounding as a response to misbehaviour.⁸ There is also somewhat of a reduction in parents giving out to the young person, though this occurs at least occasionally in the vast majority of families.

An alternative measure of family relationships and contact is whether they eat dinner together on a daily basis. Table 2.5 shows the factors associated with eating together. Model 1 shows that there is a raw difference, with Cohort '08 being slightly less likely to eat with their family daily; the percentages eating together daily are 64 per cent for Cohort '98 and 62 per cent for Cohort '08. However, Model 2 indicates that this difference is related to changes in the profile of the population. Daily dining does not vary markedly by parental education but there are clear differences by social class, with those whose parents have professional, managerial or other non-manual occupations being less likely to do so, a pattern that may be related to differences in working patterns. In keeping with these social class patterns, eating together is more common among the lowest income groups (Table A2.1). Daily dining levels are also lower among lone-parent and migrant-origin

⁸ Additional analyses (not shown here) indicate that this shift occurred across all levels of parental education.
families as well as for those living in urban areas. Rates are also lower where young people have a disability. Variation by social class does not change over time but there is an interesting shift in the gender pattern. Estimated probabilities indicate that for girls, rates decline from 65 per cent for Cohort '98 to 61 per cent for Cohort '08, while for boys, rates are broadly stable at 63 per cent. It is difficult to explain these gender differences without further exploring eating patterns and day-to-day activities among girls and boys.

TABLE 2.5LOGISTIC REGRESSION MODELS OF THE FAMILY EATING DINNER TOGETHER EVERY
DAY (ODDS RATIOS)

	Raw differences (1)	Differences controlling for family and child factors (2)	Changes in the effect of gender and education (3)	Changes in the effect of class and strain (4)
Constant	1.788	2.824	2.679	2.627
Cohort '08	0.914*	0.954	1.054	1.109
(Ref. Cohort '98)				
Female		1.016	1.102*	1.104*
(Ref. Male)				
Parental education:				
Leaving Certificate		0.940	0.926	0.942
Post-secondary		0.830**	0.844*	0.826**
Degree		0.901	0.927	0.901
(Ref. Lower secondary)				
Social class:				
Professional		0.825*	0.829*	0.967
Managerial		0.780***	0.785***	0.804*
Other non-manual		0.780***	0.786***	0.793*
Skilled manual		0.870±	0.877±	0.888
Non-employed		0.997	1.000	0.969
(Ref. Semi/unskilled manual)				
Experiencing financial strain		0.961	0.962	0.983
Parent(s) born outside Ireland		0.835**	0.836**	0.834**
Lone-parent family		0.756***	0.756***	0.759***
(Ref. Two-parent family)				
Large family		0.999	0.998	0.942
(Ref. 2 or fewer siblings)				
Disability/illness		0.897*	0.900*	0.899*
Urban location		0.781***	0.782***	0.782***
(Ref. Rural)				
Social/private rented tenure		1.059	1.052	1.052
(Ref. Own with/without mortgage)				
Female* Cohort '08			0.836*	0.832*
Leaving Certificate* Cohort '08			1.089	
Post-secondary* Cohort '08			0.974	
Degree* Cohort '08			0.950	
Professional* Cohort '08				0.745±
Managerial* Cohort '08				0.944
Non-manual* Cohort '08				0.979
Skilled* Cohort '08				0.973
Non-employed* Cohort '08				1.083
Strain* Cohort '08				0.933
Nagelkerke R ²	0.001	0.014	0.015	0.015
Ν		13,	562	

 Source:
 Growing Up in Ireland Cohorts '98 and '08.

 Note:
 *** p<.001; ** p<.01; * p<.05; + p<.10.</td>

2.3 PEER RELATIONSHIPS

Figure 2.3 shows the number of friends and of close friends reported by young people from the two cohorts. In both cases, there is a significant decline in those having large friendship groups (of six or more). For friends overall, large peer groups declined from 55 per cent to 38 per cent and from 26 per cent to 14 per cent in the case of close friends.





Source: GUI Cohort '98 and '08.

Note: C98: Cohort '98: C08: Cohort '08.

Tables 2.6 and 2.7 look at whether these changes are due to shifts in the composition of 13-year-olds and their families. Table 2.6 shows a significant decline in the size of peer groups is evident, even when individual and family background factors are taken into account. Peer groups are smaller among those with a disability, those from migrant backgrounds and among those whose parents have post-secondary or tertiary qualifications. Peer groups are larger among young people from larger families. There is little systematic variation by social class, location, income or experience of financial strain. However, those living in rented accommodation have smaller peer groups. This may relate to more frequent house moves among this group that disrupt friendship patterns (see Laurence et al., 2023). Alternatively, poorer housing quality (including accommodation being too small) in this sector may mean that young people are less likely to be able to bring friends home. Overall, girls have smaller peer groups than boys. On closer investigation, there is little difference for Cohort '98 but a much larger gender gap for Cohort '08. There is little systematic change over time in the pattern by social class, parental education or income. However, there is a tendency for the decline in peer group size to be greatest for those from non-employed households and less marked for those from higher income groups (Table A2.3).

TABLE 2.6GENERALISED LINEAR MODELS (CUMULATIVE LOGIT FUNCTION) OF NUMBER OF
FRIENDS (ODDS RATIOS)

	Raw differences (1)	Differences controlling for family and child factors (2)	Changes in the effect of gender and education (3)	Changes in the effect of class and strain (4)
Cohort '08	0.497***	0.560***	0.806*	0.750***
(Ref. Cohort '98)				
Female		0.891***	1.070	1.069
(Ref. Male)				
Parental education:				
Leaving Certificate		1.010	1.107	1.026
Post-secondary		0.845**	0.923	0.864*
Degree		0.864*	0.906	0.881*
(Ref. Lower secondary)				
Social class:				
Professional		1.002	0.996	1.035
Managerial		1.130*	1.121±	1.159±
Other non-manual		0.992	0.995	1.077
Skilled manual		0.907	0.910	0.924
Non-employed		0.966	0.970	1.162
(Ref. Semi/unskilled manual)				
Experiencing financial strain		1.018	1.016	0.971
Parent(s) born outside Ireland		0.713***	0.714***	0.714***
Lone-parent family		1.042	1.047	1.038
(Ref. Two-parent family)				
Large family		1.127**	1.128**	1.131**
(Ref. 2 or fewer siblings)				
Disability/illness		0.731***	0.733***	0.734***
Urban location		1.051	1.058	1.057
(Ref. Rural)				
Social/private rented tenure		0.859**	0.860**	0.851***
(Ref. Own with/without mortgage)				
Female* Cohort '08			0.656***	0.658***
Leaving Certificate* Cohort '08			0.730*	
Post-secondary* Cohort '08			0.829	
Degree* Cohort '08			0.898	
Professional* Cohort '08				0.928
Managerial* Cohort '08				0.946
Non-manual* Cohort '08				0.842
Skilled* Cohort '08				0.983
Non-employed* Cohort '08				0.642**
Strain* Cohort '08				1.174
Log likelihood	-11038.830	-10951.534	-10927.926	-10923.693
N		13,3	169	

Source: Growing Up in Ireland Cohorts '98 and '08.

Note: *** p<.001; ** p<.01; * p<.05; + p<.10.

The picture is fairly similar when we consider close friends rather than friends in general, with a significant decline in the size of the group over time (Table 2.7). However, there are clearer differences by some background characteristics, with smaller numbers of close friends found among migrant-origin young people and those with a disability. Those from the higher social classes (professional, managerial and other non-manual) and from the highest income group are more likely to report having more close friends (Table A2.3). As with friends in general, those from larger families tend to have more close friends while those in rented accommodation tend to have fewer close friends. There is a wider gender gap in the number of close friends among Cohort '08, with girls having fewer friends (Model 3). As with friends in general, the proportion with large networks declines more for those from jobless households than for other class groups. It would appear that the decline in the size of friendship groups resulted in less diversity by age within the group. Among Cohort '98, a fifth reported that some, most or all of their friends were two years older than them, while this had declined to just 7 per cent among Cohort '08.

GENERALISED LINEAR MODEL (CUMULATIVE LOGIT FUNCTION) OF NUMBER OF **TABLE 2.7 CLOSE FRIENDS (ODDS RATIOS)**

	Raw differences (1)	Differences controlling for family and child factors (2)	Changes in the effect of gender and education (3)	Changes in the effect of class and strain (4)
Cohort '08	0.585***	0.628***	0.720**	0.890
(Ref. Cohort '98)				
Female		0.840***	0.939	0.941
(Ref. Male)				
Parental education:				
Leaving Certificate		1.156*	1.173*	1.175**
Post-secondary		1.107±	1.153±	1.125±
Degree		1.032	1.033	1.045
(Ref. Lower secondary)				
Social class:				
Professional		1.233*	1.231*	1.418**
Managerial		1.264**	1.262***	1.359***
Other non-manual		1.159*	1.165*	1.327***
Skilled manual		0.947	0.953	1.063
Non-employed		1.095	1.101	1.379***
(Ref. Semi/unskilled manual)				
Experiencing financial strain		0.943	0.942	0.945
Parent(s) born outside Ireland		0.671***	0.672**	0.672***
Lone-parent family		1.046	1.046	1.042
(Ref. Two-parent family)				
Large family		1.115**	1.115**	1.117***
(Ref. 2 or fewer siblings)				
Disability/illness		0.689***	0.692***	0.690***
Urban location		0.939±	0.941±	0.944±
(Ref. Rural)				
Social/private rented tenure		0.845***	0.843***	0.839***
(Ref. Own with/without mortgage)				
Female* Cohort '08			0.776***	0.777***
Leaving Certificate* Cohort '08			0.971	
Post-secondary* Cohort '08			0.946	
Degree* Cohort '08			1.020	
Professional* Cohort '08				0.750
Managerial* Cohort '08				0.861
Non-manual* Cohort '08				0.753*
Skilled* Cohort '08				0.787±
Non-employed* Cohort '08				0.584***
Strain* Cohort '08				0.983
Log likelihood	-11052.313	-10910.521	-10902.760	-10894.535
N		13,0	091	

 Source:
 Growing Up in Ireland Cohorts '98 and '08.

 Note:
 *** p<.001; ** p<.01; * p<.05; + p<.10.</td>

Because of the absence of other measures of friendship quality (see Chapter 1), the SDQ peer problems subscale is used as a proxy. This scale varies from 0 to 10, with higher scores indicating greater problems, with a mean of 1.3 and a standard deviation of 1.6. As noted in Chapter 1, this measure will capture having few friends as well as the quality of interaction with them. Table 2.8 shows a significant increase over time in the level of peer problems (Model 1). Model 2 indicates that most (almost two-thirds) of this increase is accounted for by a change in the profile of young people, namely, an increase in the proportion with a disability, from migrantorigin families and those living in rented accommodation, all of which are associated with greater peer difficulties. The differences by disability and rental tenure are large, at more than half a standard deviation. Gender differences are very modest, with only slightly fewer problems found among girls. Peer difficulties do not vary markedly by education but are less evident in professional/managerial and higher-income families (Tables 2.8 and A2.2). Such difficulties are also more prevalent in families experiencing financial strain and those living in urban areas while they are somewhat less evident in larger families. There is little change over time in the patterning of peer relations by gender or social background, though those living in non-employed households are more likely than other groups to have had an increase in peer difficulties.



FIGURE 2.4 MEAN SDQ PEER PROBLEMS BY NUMBER OF FRIENDS AND OF CLOSE FRIENDS BY COHORT

Source: GUI Cohort '98 and '08.

Note: For Cohort '98, levels of peer problems do not differ between the largest two groups but all other pair-wise comparisons are significant. For Cohort '08, the main difference for number of friends is between the lowest group and all others. For close friends, levels of peer problems differ significantly between all groups.

Figure 2.4 shows that there is a marked relationship between the number of friends a young person has and the extent of mother-reported difficulties interacting with their peers. Peer difficulties are particularly evident for those with very small friendship groups and decline with size of the peer group.

|--|

	Raw differences (1)	Differences controlling for family and child factors (2)	Changes in the effect of gender and education (3)	Changes in the effect of class and strain (4)
Constant	1.081	0.990	0.961	1.019
Cohort '08	0.167***	0.058*	0.197	0.016
(Ref. Cohort '98)				
Female		-0.047±	-0.073*	-0.074*
(Ref. Male)				
Parental education:				
Leaving Certificate		-0.072	-0.051	-0.079
Post-secondary		-0.031	0.015	-0.045
Degree		-0.040	0.016	-0.050
(Ref. Lower secondary)				
Social class:				
Professional		-0.164**	-0.162**	-0.163*
Managerial		-0.151**	-0.149**	-0.141*
Other non-manual		-0.081	-0.079	-0.093
Skilled manual		-0.048	-0.075	-0.115
Non-employed		0.191**	0.185*	0.064
(Ref. Semi/unskilled manual)				
Experiencing financial strain		0.194***	0.197***	0.210***
Parent(s) born outside Ireland		0.184***	0.183***	0.180***
Lone-parent family		0.075±	0.074±	0.079±
(Ref. Two-parent family)				
Large family		-0.094**	-0.094**	-0.095**
(Ref. 2 or fewer siblings)				
Disability/illness		0.839***	0.837***	0.837***
Urban location		0.107***	0.108***	0.107***
(Ref. Rural)				
Social/private rented tenure		0.961***	0.200***	0.203***
(Ref. Own with/without mortgage)				
Female* Cohort '08			0.060	0.059
Leaving Certificate* Cohort '08			-0.045	
Post-secondary* Cohort '08			-0.178	
Degree* Cohort '08			-0.201	
Professional* Cohort '08				-0.002
Managerial* Cohort '08				-0.022
Non-manual* Cohort '08				0.026
Skilled* Cohort '08				0.091
Non-employed* Cohort '08				0.271±
Strain* Cohort '08				-0.048
Adjusted R ²	0.003	0.081	0.081	0.081
Ν		13,8	872	

 Source:
 Growing Up in Ireland Cohorts '98 and '08.

 Note:
 *** p<.001; ** p<.01; * p<.05; + p<.10.</th>

2.4 SUMMARY

This chapter has explored the nature of relationships with parents and peers among 13-year-olds and whether these relationships have changed over time. The findings point to an improvement in the quality of parent-child relationships, with a reduction in conflict and improved responsiveness. How parents deal with misbehaviour has become more democratic, with greater emphasis on explaining what the young person did wrong and less reliance on shouting at them or more punitive measures such as grounding them. The regularity of families having dinner together is broadly stable when we take account of the shift in profile of 13-year-olds and their families over time.

The findings show a change over time in the size of peer groups, with a marked reduction in the number of friends overall and of close friends reported by young people. A greater reduction was evident for girls than boys. The shift was accompanied by less age diversity in the peer group, with Cohort '08 participants less likely to report socialising with much older peers. A potential explanation lies in the effects of the pandemic. While the vast majority of 13-year-olds reported always being able to keep in touch with their friends during the most recent period of school closures, those who were not had smaller friendship groups subsequently and were less likely to 'hang out' with their friends frequently. Unfortunately, changes over time in the questions used means that there is no comparable measure of the quality of friendships. The mother-reported SDQ peer difficulties subscale is used as a proxy. This measure indicates an increase in problems interacting with peers over time, an increase that is largely (but not wholly) driven by the increase in young people of migrant origin, with a disability or living in rented accommodation, all factors associated with poorer peer relations. Quantity and quality are intertwined, with those with very small friendship groups having greater difficulties interacting with others.

TABLE A2.1SENSITIVITY ANALYSES OF PARENT-CHILD RELATIONSHIPS (PARENT-REPORTED)TO INCLUDE HOUSEHOLD INCOME QUINTILE

	Maternal conflict	Paternal conflict	Gets on very well with mother (Odds ratio)	Maternal responsiveness	Eating together daily (Odds ratios)
		13.840	1.719	20.134	2.446
Cohort '08	-2.472***	-2.681***	1.482**	2.101***	1.065
(Ref. Cohort '98)					
Household income quintile:					
Quintile 2	0.013	0.773**	-0.207*	0.233±	0.973
Quintile 3	-0.023	0.194	-0.046	0.253±	0.658***
Quintile 4	0.017	0.122	-0.031	0.605***	0.842*
Quintile 5	-0.128	0.476±	-0.268**	0.412**	0.744***
Income missing	0.141	0.275	0.840	0.227	0.929
(Ref. Lowest quintile)					
Quintile 2* Cohort '08	0.214	-0.801±	1.052	-0.445±	0.899
Quintile 3* Cohort '08	0.885*	0.695	0.536**	-0.813**	1.218
Quintile 4* Cohort '08	0.358	0.653	0.591**	-1.435***	0.812±
Quintile 5* Cohort '08	1.055**	0.668	0.919	-0.781**	0.936
Income missing* Cohort '08	-0.514	0.043	0.846	-0.926**	0.857
Adjusted/Nagelkerke R ²	0.059	0.053	0.025	0.028	0.017
Ν	14,158	10,447	10,109	9,881	13,562

Source: Growing Up in Ireland Cohorts '98 and '08.

Note: *** p<.001; ** p<.01; * p<.05; + p<.10. The models also control for gender, family structure, family size, migrant status, urban/rural location and housing tenure.

TABLE A2.2 OLS REGRESSION MODELS OF RELATIONSHIP BETWEEN SOCIO-EMOTIONAL WELLBEING AND PARENT-CHILD CONFLICT WELLBEING WELLBEING

	Materna	al conflict	Paternal conflict		
	Model 1 Model 2		Model 1	Model 2	
Constant	15.231 13.186		14.915	13.810	
Cohort '08	-1.579*** -2.049***		-2.128***	-2.331***	
Female	0.015 -0.435**		0.256	-0.062	
Female* Cohort '08	0.916*** 0.598***		0.554*	0.302	
SDQ emotional problems		1.191***		0.720***	
Adjusted R ²	0.009 0.163		0.027	0.092	
Ν	13,	562	10,447		

Source: Growing Up in Ireland Cohorts '98 and '08.

Note: *** p<.001; ** p<.01; * p<.05; + p<.10.

Text	Number of friends (Odds ratios)	Number of close friends (Odds ratios)	SDQ peer problems
Constant			0.976
Cohort '08	0.564***	0.611***	-0.010
(Ref. Cohort '98)			
Household income quintile:			
Quintile 2	0.969	0.972	-0.069
Quintile 3	0.935	1.031	-0.083
Quintile 4	0.930	1.039	-0.130*
Quintile 5	1.013	1.185*	-0.127*
Income missing	1.123	1.168	-0.219**
(Ref. Lowest quintile)			
Quintile 2* Cohort '08	1.208±	1.303*	0.044
Quintile 3* Cohort '08	1.224±	1.140	-0.022
Quintile 4* Cohort '08	1.251*	1.277*	0.012
Quintile 5* Cohort '08	1.259*	1.200	-0.032
Income missing* Cohort '08	1.049	1.029	0.197±
Adjusted R ²	-	-	0.077
Log likelihood	-10939.702	-10913.232	-
Ν	13,169	13,091	13,872

SENSITIVITY ANALYSES OF PEER GROUP TO INCLUDE HOUSEHOLD INCOME QUINTILE TABLE A2.3

Source:

Growing Up in Ireland Cohorts '98 and '08. *** p<.001; ** p<.01; * p<.05; + p<.10. The models also control for gender, family structure, family size, migrant Note: status, urban/rural location and housing tenure.

CHAPTER 3

Day-to-day activities

3.1 INTRODUCTION

This chapter turns attention to the day-to-day activities engaged in by 13-yearolds. Section 3.2 looks at involvement in organised sport and in levels of physical exercise. It also looks at involvement in cultural activities (such as music or drama classes) and in organised clubs (such as youth clubs or Scouts/Guides). Section 3.3 examines three types of screen time: watching television, playing video/computer games and other screen use. As in Chapter 2, multivariate models are used to look at changes over time overall and by gender and social background.

3.2 STRUCTURED ACTIVITIES AND PHYSICAL EXERCISE

Because of changes in the response categories used (see Chapter 1), involvement in organised sport distinguishes between those who were involved at least once a week and others. Using this measure, levels of involvement are found to have increased over time, from 65 per cent for Cohort '98 to 70 per cent for Cohort '08. As discussed in Chapter 1, fieldwork for Cohort '08 took place over a period when there had been easing of pandemic-related restrictions but not all restrictions were removed. To capture this, young people who did not take part in structured activities at the time of the survey were asked whether this was related to lack of interest, pandemic restrictions or another reason. For sport, 1.7 per cent of the total cohort were not taking part in sport and attributed this to pandemic restrictions. Thus, the increase over time in participation may be slightly underestimated because of the survey timing.

Table 3.1 (Model 2) indicates that the increase in weekly involvement in organised sport is not explained by changes in the composition of the adolescent group. Girls are much less likely to be involved in organised sport than boys. Involvement increased for both girls and boys but the size of the gender gap did not change over time. A clear social gradient is evident, with the highest levels of involvement among those from professional or graduate families. Even taking account of parental education and social class, lower levels of involvement are found among those in rented accommodation. This may relate to involvement in organised activities being disrupted by house moves (see Laurence et al., 2023) and/or the lack of green space or local facilities in certain areas. The scale of the gender and social gaps is evident from Figure 3.1. There are also clear gradients by household income (not illustrated, see Table A3.1). The interaction terms (Models 3 and 4) indicate that the effects of gender, social class, parental education or income did not change over time. In contrast, the effects of financial strain did change. Levels of involvement are lower among those whose families report difficulty making ends

meet. However, unlike other groups, there was no increase in involvement over time for this group (see Figure 3.1), meaning that the gap by financial strain is greater among Cohort '08 members. Other measures of individual and family circumstances are strongly associated with sports involvement. Levels of involvement are much lower among those with a disability and among those from migrant-origin families.





Source: Derived from Models 2 and 4 in Table 3.1.

TABLE 3.1LOGISTIC REGRESSION MODELS OF THE 13-YEAR-OLD TAKING PART IN ORGANISED
SPORTS AT LEAST WEEKLY (ODDS RATIOS)

	Raw differences (1)	Differences controlling for family and child factors (2)	Changes in the effect of gender and education (3)	Changes in the effect of class and strain (4)
Constant	1.831	1.847	1.791	1.709
Cohort '08	1.270***	1.325***	1.443**	1.575***
(Ref. Cohort '98)				
Female		0.574***	0.585***	0.582***
(Ref. Male)				
Parental education:				
Leaving Certificate		1.310***	1.271**	1.319***
Post-secondary		1.456***	1.556***	1.457***
Degree		2.061***	2.177***	2.078***
(Ref. Lower secondary)				
Social class:				
Professional		1.954***	1.965***	2.015***
Managerial		1.466***	1.473***	1.569***
Other non-manual		1.302***	1.313***	1.426***
Skilled manual		1.076	1.083	1.043
Non-employed		0.861±	0.860±	0.937
(Ref. Semi/unskilled manual)				
Experiencing financial strain		0.862**	0.864**	0.937
Parent(s) born outside Ireland		0.599***	0.598***	0.592***
Lone-parent family		0.909±	0.908±	0.912
(Ref. Two-parent family)				
Large family		1.039	1.039	1.045
(Ref. 2 or fewer siblings)				
Disability/illness		0.540***	0.540***	0.539***
Urban location		0.930±	0.931±	0.930±
(Ref. Rural)				
Social/private rented tenure		0.707***	0.702***	0.705***
(Ref. Own with/without mortgage)				
Female* Cohort '08			0.966	0.962
Leaving Certificate* Cohort '08			1.184	
Post-secondary* Cohort '08			0.861	
Degree* Cohort '08			0.865	
Professional* Cohort '08				0.913
Managerial* Cohort '08				0.846
Non-manual*Cohort '08				0.812
Skilled* Cohort '08				1.099
Non-employed* Cohort '08				0.822
Strain* Cohort '08				0.743*
Nagelkerke R ²	0.004	0.133	0.134	0.135
Ν		13.2	236	

 Source:
 Growing Up in Ireland Cohorts '98 and '08.

 Note:
 *** p<.001; ** p<.01; * p<.05; + p<.10.</td>

Because of changes in the categories used, analyses of levels of exercise (hard and light) focus on the factors associated with having low levels of involvement (the equivalent of two days or less per fortnight, see Chapter 1). While these are more comparable, it is worth noting some caveats. First, the change in timeframe from 14 to 7 days may have altered respondent recall and/or reporting. Second, more examples were given for hard exercise for Cohort '08 and some of the examples were potentially more inclusive of a broader range of interests (such as dancing and swimming). Furthermore, some public health restrictions were still in place at the time of the survey so exercise may have been a more viable activity than other activities.

Having low levels of hard exercise is much less common for Cohort '08, a difference that is not explained by changes in their profile (Models 1 and 2, Table 3.2). Girls have much lower levels of hard exercise than boys, though the interaction term in Model 3 indicates that the gender gap closes somewhat over time but remains substantial. There is a clear gradient by parental education with those whose parents have only secondary education being more likely to have low levels, a pattern that does not change over time. In contrast, there is a slight reduction in the level of social class inequality, though those from manual or non-employed households continue to have lower levels of hard exercise than other social class groups. Involvement also varies by income, with the highest levels found in the highest income quintiles (Table A3.1). Over and above parental education and social class, being from a migrant background is associated with lower levels of hard exercise as is living in rented accommodation. The latter pattern may relate to access to green space and the presence of local facilities and merits further investigation. Those with a disability have much lower levels of hard exercise than their peers.



FIGURE 3.2 ACTUAL PERCENTAGES WITH LOW LEVELS OF HARD EXERCISE BY SOCIAL CLASS

TABLE 3.2LOGISTIC REGRESSION MODELS OF THE 13-YEAR-OLD HAVING LOW LEVELS OF
INVOLVEMENT IN HARD EXERCISE (ODDS RATIOS)

	Raw differences (1)	Differences controlling for family and child factors (2)	Changes in the effect of gender and education (3)	Changes in the effect of class and strain (4)
Constant	0.316	0.267	0.259	0.280
Cohort '08	0.320***	0.274***	0.292***	0.228***
(Ref. Cohort '98)				
Female		2.200***	2.358***	2.366***
(Ref. Male)				
Parental education:				
Leaving Certificate		0.854*	0.906	0.856*
Post-secondary		0.761***	0.704***	0.770**
Degree		0.668***	0.636***	0.674***
(Ref. Lower secondary)				
Social class:				
Professional		0.658***	0.653***	0.550***
Managerial		0.643***	0.639***	0.586***
Other non-manual		0.642***	0.634***	0.563***
Skilled manual		0.851±	0.846±	0.778*
Non-employed		1.005	1.010	0.816±
(Ref. Semi/unskilled manual)				
Experiencing financial strain		0.995	0.990	1.048
Parent(s) born outside Ireland		1.534***	1.548***	1.553***
Lone-parent family		1.018	1.023	1.021
(Ref. Two-parent family)				
Large family		0.906	0.905	0.898±
(Ref. 2 or fewer siblings)				
Disability/illness		1.993***	2.001***	2.009***
Urban location		1.036	1.035	1.033
(Ref. Rural)				
Social/private rented tenure		1.319***	1.331***	1.325***
(Ref. Own with/without mortgage)				
Female* Cohort '08			0.759*	0.750*
Leaving Certificate* Cohort '08			0.581*	
Post-secondary* Cohort '08			1.308	
Degree* Cohort '08			1.276	
Professional* Cohort '08				1.788*
Managerial* Cohort '08				1.378±
Non-manual* Cohort '08				1.630*
Skilled* Cohort '08				1.389
Non-employed* Cohort '08				2.089***
Strain* Cohort '08				0.705±
Nagelkerke R ²	0.064	0.148	0.151	0.151
Ν		13,	221	

 Source:
 Growing Up in Ireland Cohorts '98 and '08.

 Note:
 *** p<.001; ** p<.01; * p<.05; + p<.10.</td>

Patterns of involvement in light exercise are broadly similar to those for hard exercise, with a significant improvement in levels over time (Table 3.3). Unlike hard exercise, however, there is no evidence of a gender gap. Light exercise levels are lower among those whose parents have secondary education, a gap that narrows somewhat over time. Those in the highest income group have higher levels of light exercise but there is some narrowing of the income gap over time. Levels of light exercise are much lower among those with a disability.

TABLE 3.3LOGISTIC REGRESSION MODELS OF THE 13-YEAR-OLD HAVING LOW LEVELS OF
INVOLVEMENT IN LIGHT EXERCISE (ODDS RATIOS)

	Raw differences (1)	Differences controlling for family and child factors (2)	Changes in the effect of gender and education (3)	Changes in the effect of class and strain (4)
Constant	0.658	0.792	0.829	0.878
Cohort '08	0.186***	0.195***	0.156***	0.125***
(Ref. Cohort '98)				
Female		0.984	0.979	0.982
(Ref. Male)				
Parental education:				
Leaving Certificate		0.968	0.949	0.965
Post-secondary		0.808**	0.857±	0.827*
Degree		0.704***	0.621***	0.705***
(Ref. Lower secondary)				
Social class:				
Professional		0.882	0.857	0.769*
Managerial		0.812**	0.790**	0.689***
Other non-manual		0.878	0.863±	0.837±
Skilled manual		0.938	0.927	0.865
Non-employed		0.956	0.948	0.802*
(Ref. Semi/unskilled manual)				
Experiencing financial strain		1.095±	1.090	1.089
Parent(s) born outside Ireland		0.971	0.972	0.980
Lone-parent family		1.113±	1.109±	1.115±
(Ref. Two-parent family)				
Large family		1.106±	1.108*	1.102±
(Ref. 2 or fewer siblings)				
Disability/illness		1.195***	1.199***	1.202***
Urban location		0.973	0.971	0.973
(Ref. Rural)				
Social/private rented tenure		1.086	1.098	1.096
(Ref. Own with/without mortgage)				
Female* Cohort '08			1.003	1.007
Leaving Certificate* Cohort '08			1.122	
Post-secondary* Cohort '08			0.971	
Degree* Cohort '08			1.757**	
Professional* Cohort '08				1.706*
Managerial* Cohort '08				1.945***
Non-manual* Cohort '08				1.155
Skilled* Cohort '08				1.349
Non-employed* Cohort '08				1.963***
Strain* Cohort '08				1.043
Nagelkerke R ²	0.152	0.166	0.169	0.168
Ν		13,2	220	

Weekly involvement in structured cultural activities increased significantly over time, from 29 per cent in Cohort '98 to 36 per cent in Cohort '08.⁹ It should be noted that the explicit mention of arts and crafts in the Cohort '08 survey may have increased the figure slightly. Nonetheless, there is no evidence of the slight decline in participation found among these young people at the age of nine. This overall picture conceals considerable complexity by gender and social background. As at nine years of age, gender makes the largest difference to cultural participation, with much higher levels found among girls (Table 3.4). The interaction term indicates that the gender gap narrowed slightly over time but remained nonetheless substantial. There is a clear social gradient in cultural participation, with the greatest involvement found among those whose parents are graduates or in professional occupations and those whose families have higher levels of income. There is no shift over time in the education or class gap. However, there is some evidence of a slight narrowing of the gap between the top income group and others, though again the difference remains large. Young people living in urban areas are much more likely to be involved in structured cultural activities but few other background factors influence participation.

In contrast, weekly participation in organised groups (such as youth clubs or Scouts/Guides) declined from 21 per cent for Cohort '98 to 16 per cent for Cohort '08. This decline is not accounted for by changes in the profile of the group (Table 3.5, Model 2) but may be, at least partly, due to pandemic restrictions as 3.7 per cent of the total said they were not involved for this reason. There are no gender differences in participation and very little systematic variation by social background, except for lower involvement among those from jobless households. There is some shift by social class over time (Model 4), with a decline among all social class groups except those from semi/unskilled backgrounds. Levels of involvement are much higher among those living in rented accommodation and, to some extent, among those from larger families. Participation is found to be much lower in urban than in rural areas.

⁹ As with sports, some cultural activities may have been curtailed due to pandemic restrictions. The proportion of the total cohort who indicated they were not involved in cultural activities because of restrictions was 2.4 per cent so the figure for involvement may be an underestimate.

TABLE 3.4LOGISTIC REGRESSION MODELS OF THE 13-YEAR-OLD HAVING AT LEAST WEEKLY
INVOLVEMENT IN CULTURAL ACTIVITIES (ODDS RATIOS)

	Raw differences (1)	Differences controlling for family and child factors (2)	Changes in the effect of gender and education (3)	Changes in the effect of class and strain (4)
Constant	0.317	0.196	0.099	0.100
Cohort '08	1.409***	1.263***	2.037***	1.688***
(Ref. Cohort '98)				
Female		4.063***	4.859***	4.885***
(Ref. Male)				
Parental education:				
Leaving Certificate		1.008	1.039	0.991
Post-secondary		1.353**	1.449**	1.298*
Degree		2.034***	2.284***	1.984***
(Ref. Lower secondary)				
Social class:				
Professional		1.240*	1.246*	1.420**
Managerial		1.108	1.111	1.203
Other non-manual		1.102	1.113	1.053
Skilled manual		0.968	0.974	0.925
Non-employed		0.896	0.895	0.634*
(Ref. Semi/unskilled manual)				
Experiencing financial strain		0.883±	0.883±	0.893
Parent(s) born outside Ireland		0.926	0.928	0.914
Lone-parent family		0.908	0.904	1.090±
(Ref. Two-parent family)				
Large family		0.918±	0.917±	1.142**
(Ref. 2 or fewer siblings)				
Disability/illness		1.087±	1.090±	1.090
Urban location		1.140**	1.143**	1.142**
(Ref. Rural)				
Social/private rented tenure		0.933	0.924	0.922
(Ref. Own with/without mortgage)				
Female* Cohort '08			0.707***	0.698***
Leaving Certificate* Cohort '08			0.954	
Post-secondary* Cohort '08			0.787	
Degree* Cohort '08			0.721	
Professional* Cohort '08				0.784
Managerial* Cohort '08				0.844
Non-manual* Cohort '08				1.123
Skilled* Cohort '08				1.126
Non-employed* Cohort '08				1.939**
Strain* Cohort '08				0.977
Nagelkerke R ²	0.009	0.159	0.161	0.163
Ν		13.	509	

TABLE 3.5LOGISTIC REGRESSION MODELS OF THE 13-YEAR-OLD HAVING AT LEAST WEEKLY
INVOLVEMENT IN ORGANISED CLUBS OR GROUPS (ODDS RATIOS)

	Raw differences (1)	Differences controlling for family and child factors (2)	Changes in the effect of gender and education (3)	Changes in the effect of class and strain (4)
Constant	0.270	0.328	0.315	0.272
Cohort '08	0.690***	0.682***	0.774±	1.024
(Ref. Cohort '98)				
Female		0.976	1.002	1.007
(Ref. Male)				
Parental education:				
Leaving Certificate		0.833*	0.884	0.828*
Post-secondary		0.870±	0.947	0.984
Degree		0.944	0.928	0.954
(Ref. Lower secondary)				
Social class:				
Professional		1.042	1.030	1.113
Managerial		0.950	0.939	1.099
Other non-manual		0.988	0.982	1.289*
Skilled manual		0.984	0.880	1.137
Non-employed		0.807*	0.804*	0.975
(Ref. Semi/unskilled manual)				
Experiencing financial strain		1.083	1.080	1.057
Parent(s) born outside Ireland		0.993	0.991	0.995
Lone-parent family		0.916	0.916	0.919
(Ref. Two-parent family)				
Large family		1.126*	1.127*	1.130*
(Ref. 2 or fewer siblings)				
Disability/illness		0.983	0.984	0.975
Urban location		0.795***	0.797***	0.802***
(Ref. Rural)				
Social/private rented tenure		1.312***	1.323***	1.308***
(Ref. Own with/without mortgage)				
Female* Cohort '08			0.934	0.939
Leaving Certificate* Cohort '08			0.767	
Post-secondary* Cohort '08			0.824	
Degree* Cohort '08			1.025	
Professional* Cohort '08				0.829
Managerial* Cohort '08				0.713*
Non-manual* Cohort '08				0.513***
Skilled* Cohort '08				0.528***
Non-employed* Cohort '08				0.634*
Strain* Cohort '08				1.128
Nagelkerke R ²	0.008	0.016	0.017	0.019
Ν		13.2	235	



FIGURE 3.3 LEVEL OF INVOLVEMENT IN ORGANISED ACTIVITIES (SPORTS, CULTURAL AND CLUBS) BY COHORT

Source: GUI Cohorts '98 and '08.

Looking at involvement across the three types of structured activities (organised sport, cultural activities and clubs such as youth clubs) shows that the majority of 13-year-olds across both cohorts are involved in at least one such activity (Figure 3.3). Between a fifth and a quarter have no such involvement, with the most common form of involvement being one activity. Between cohorts, there is an increase in the proportion involved in at least one structured activity.



FIGURE 3.4 TIME SPENT READING FOR PLEASURE BY SOCIAL CLASS AND GENDER, COHORT '98 (ACTUAL PERCENTAGES)



FIGURE 3.5 FREQUENCY OF READING FOR PLEASURE BY SOCIAL CLASS AND GENDER, COHORT '08 (ACTUAL PERCENTAGES)



Analyses of the two cohorts at age nine had indicated a decline in the frequency of reading for pleasure over time (Smyth, 2022). Unfortunately, it is not possible to make a similar comparison at 13 years of age because of differences in the question wording – focusing on time per day for Cohort '98 and number of days per week for Cohort '08. Nonetheless, a descriptive analysis of patterns by gender and social background reveals interesting insights. For Cohort '98, girls were much more likely than boys to spend more than an hour a day reading, with this gender difference significant in all social classes except the professional group (Figure 3.4). Among boys, there was a clear social gradient in reading patterns, with a sizeable proportion – a third – of boys from working-class or non-employed households spending no time reading. There is much less variation by social class among girls, though the proportion not spending any time reading is higher among the other non-manual and working-class groups than among the professional/managerial classes.

Among Cohort '08, significant gender differences in reading frequency are found within all social classes (Figure 3.5). As for Cohort '98, social class differences are stronger among boys than girls. While the answer categories cannot be compared across cohorts, taken in conjunction with comparable findings for the two cohorts at age nine, there does seem to be tentative evidence of an increase in those who hardly ever read for pleasure. Around half of boys in the other non-manual and working-class groups report reading for pleasure less than once a week or never and this figure is almost four in ten even for those from the professional/managerial classes.

3.3 SCREEN TIME

In both cohorts, young people were asked about the time they spent watching television (including on another device), playing video/computer games and other online or screen-based activities on a regular term time day. As discussed in Chapter 1, there may be increased blurring between these categories of activities over time. Respondents were explicitly told not to count in-school screen time. Further research is needed to explore the type of screen use (though the categories vary over time), including for educational purposes.¹⁰

Figure 3.6 shows the time spent on these different activities between cohorts. Very significant changes are evident, with a sizeable reduction in the time spent watching television and in playing video/computer games, and a large shift to other screen-based activities.



FIGURE 3.6 TIME SPENT ON DIFFERENT TYPES OF SCREEN-BASED ACTIVITIES BY COHORT

Source: GUI Cohorts '98 and '08.

Looking at TV watching, there was a marked reduction in longer periods (four or more hours) from almost a third of those in Cohort '98 to just over a tenth for Cohort '08. The majority of Cohort '08 watch TV for two or fewer hours while this was the case for only less than half of Cohort '98. The analyses presented in Table 3.5 indicate this reduction was not accounted for by changes in the profile of 13-year-olds. Girls watch slightly less TV than boys, with this difference remaining constant over time. TV watching varies markedly by family background, being lowest for those with graduate or professional parents and higher for those living

¹⁰ Preliminary analysis would appear to suggest that schoolwork plays a modest role in these trends over time. Half of Cohort '08 reported never or hardly ever using their smartphone or tablet to do homework while two-thirds of Cohort '98 reported having used the internet for homework.

in lone-parent families. Experience of financial strain is significantly associated with more TV time as is living in rented accommodation and living in an urban area.





Some differences over time were evident in the social patterning of TV watching. Figure 3.7 shows a clear gradient by parental education in watching TV for two or fewer hours per day among both cohorts. However, the difference is somewhat less among Cohort '08 families, meaning that TV time among graduate families has not declined to the same extent as for other families. The interaction term for social class reveals a similar pattern, with (already low) levels of TV watching among professional families declining less than for other groups. In contrast, the greater tendency of those experiencing financial strain to watch more TV is greater among Cohort '08 members.

Source: Derived from Model 3 in Table 3.6.

TABLE 3.6GENERALISED LINEAR MODELS (CUMULATIVE LOGIT FUNCTION) OF TIME SPENT
WATCHING TELEVISION (ODDS RATIOS)

	Raw differences (1)	Differences controlling for family and child factors (2)	Changes in the effect of gender and education (3)	Changes in the effect of class and strain (4)
Cohort '08	0.280***	0.296***	0.236***	0.252***
(Ref. Cohort '98)				
Female		0.902**	0.883***	0.885**
(Ref. Male)				
Parental education:				
Leaving Certificate		0.844**	0.799**	0.838**
Post-secondary		0.770***	0.743***	0.781***
Degree		0.640***	0.564***	0.643***
(Ref. Lower secondary)				
Social class:				
Professional		0.711***	0.703***	0.561***
Managerial		0.891±	0.882*	0.833*
Other non-manual		0.908	0.899±	0.932
Skilled manual		0.879*	0.874*	0.853±
Non-employed		0.917	0.918	0.881
(Ref. Semi/unskilled manual)				
Experiencing financial strain		1.215***	1.210***	1.130*
Parent(s) born outside Ireland		1.055	1.058	1.060
Lone-parent family		1.096±	1.092±	1.092±
(Ref. Two-parent family)				
Large family		0.931±	0.932±	0.934±
(Ref. 2 or fewer siblings)				
Disability/illness		1.043	1.046	1.045
Urban location		1.101**	1.097**	1.101**
(Ref. Rural)				
Social/private rented tenure		1.119*	1.124*	1.123*
(Ref. Own with/without mortgage)				
Female* Cohort '08			1.038	1.048
Leaving Certificate* Cohort '08			1.179	
Post-secondary* Cohort '08			1.171	
Degree* Cohort '08			1.396**	
Professional* Cohort '08				1.587**
Managerial* Cohort '08				1.161
Non-manual* Cohort '08				0.930
Skilled* Cohort '08				1.052
Non-employed* Cohort '08				1.089
Strain* Cohort '08				1.334**
Log likelihood		-15103.947	-10945.581	-10938.426
Ν		12,	700	

 Source:
 Growing Up in Ireland Cohorts '98 and '08.

 Note:
 *** p<.001; ** p<.01; * p<.05; + p<.10.</td>

There was a very large decline in the proportion of 13-year-olds spending time playing video/computer games, with the proportion spending no time increasing from 10 per cent to 40 per cent over the decade (Figure 3.6). At the other end of the spectrum, the proportion spending four or more hours on computer games declined from 17 to 6 per cent. Very large gender differences are evident, with much lower levels of involvement among girls than boys (Table 3.7, Model 2). However, over time the decline was somewhat less in relative terms for girls than boys (see interaction term in Model 3). Computer gaming levels were much lower among those from graduate or professional families and this pattern did not change over time. Levels were also lower among those for higher-income families, though the relative decline over time was somewhat less for those group. Levels of involvement were greater among those in urban areas and those living in rented accommodation; they were also higher among those with a disability and those from migrant-origin and lone-parent families. Gaming levels were lower among those from larger families.

TABLE 3.7GENERALISED LINEAR MODELS (CUMULATIVE LOGIT FUNCTION) OF TIME SPENT
PLAYING VIDEO/COMPUTER GAMES (ODDS RATIOS)

	Raw differences (1)	Differences controlling for family and child factors (2)	Changes in the effect of gender and education (3)	Changes in the effect of class and strain (4)
Cohort '08	0.279***	0.276***	0.587***	0.757**
(Ref. Cohort '98)				
Female		0.601***	1.354***	1.358***
(Ref. Male)				
Parental education:				
Leaving Certificate		0.827***	0.817**	0.863*
Post-secondary		0.734***	0.742***	0.812***
Degree		0.582***	0.615***	0.623***
(Ref. Lower secondary)				
Social class:				
Professional		0.673***	0.672***	0.742**
Managerial		0.808***	0.821***	0.838*
Other non-manual		0.811***	0.846**	0.884
Skilled manual		0.808***	0.847**	0.963
Non-employed		0.847*	0.860*	0.908
(Ref. Semi/unskilled manual)				
Experiencing financial strain		1.012	1.000	0.997
Parent(s) born outside Ireland		1.150*	1.210**	1.217***
Lone-parent family		1.095±	1.125*	1.127*
(Ref. Two-parent family)				
Large family		0.858***	0.848***	0.848***
(Ref. 2 or fewer siblings)				
Disability/illness		1.131***	1.171***	1.166***
Urban location		1.160***	1.189***	1.195***
(Ref. Rural)				
Social/private rented tenure		1.181***	1.154**	1.162**
(Ref. Own with/without mortgage)				
Female* Cohort '08			0.128***	0.128****
Leaving Certificate* Cohort '08			1.218	
Post-secondary* Cohort '08			1.228±	
Degree* Cohort '08			1.069	
Professional* Cohort '08				0.800
Managerial* Cohort '08				0.942
Non-manual* Cohort '08				0.897
Skilled* Cohort '08				0.712**
Non-employed* Cohort '08				0.869
Strain* Cohort '08				1.019
Log likelihood		-15049,154	-10562,862	-10560.215
N		12.7	726	

 Source:
 Growing Up in Ireland Cohorts '98 and '08.

 Note:
 *** p<.001; ** p<.01; * p<.05; + p<.10.</td>

There was a very significant shift over time in other screen time (Figure 3.6). Among Cohort '98, almost half (46%) of 13-year-olds did not engage in screen time (other than TV or video/computer games) while this was the case for less than a tenth of Cohort '08 members. At the other end of the spectrum, spending three or more hours on screens increased from 15 per cent to 31 per cent. Table 3.8 indicates that this shift was not explained by changes in the profile of adolescents and their families. The results show a very significant shift in the gender patterning of screen time, with girls spending less time online than boys among Cohort '98 but a reversal of this pattern by Cohort '08. Descriptive analyses for Cohort '08 indicate that 38 per cent of girls spend two or more hours online compared with a quarter of boys.





Source: Derived from Model 3 in Table 3.8.

A shift over time by social background is also evident. Young people with graduate parents spend less time online than other groups but the relative gap narrows somewhat over time (Figure 3.8). Similarly, the gap by social class and income also appears to narrow over time (Model 4, Table 3.8 and Table A3.2). These patterns are consistent with those found for mobile phone ownership at the age of nine; early adopters in Cohort '98 were more likely to be from less advantaged groups but this social gap diminished over time. Screen time is somewhat higher in urban areas and lower for those from larger families but there is systematic variation by family type, migrant origin or having a disability.

Table 3.9 looks at the relationship between screen time and involvement in other activities. More time spent watching TV is associated with less involvement in organised sports, cultural activities and clubs, and greater likelihood of low levels of engagement in hard and light exercise. However, the thresholds vary across

outcomes. For example, only those who spend a lot of time – four or more hours – watching TV are less involved in clubs. In contrast, anything more than an hour's viewing is linked to lower levels of exercise and less cultural engagement. High levels of gaming are similarly associated with lower sports and cultural engagement and involvement in clubs. However, gaming time is linked to levels of hard exercise but not significantly related to light exercise.

TABLE 3.8GENERALISED LINEAR MODELS (CUMULATIVE LOGIT FUNCTION) OF OTHER
SCREEN TIME (ODDS RATIOS)

	Raw differences	Differences controlling for family and child factors	Changes in the effect of gender and education	Changes in the effect of class and strain
Cohort '08	3.679***	3.871***	0.540***	0.360***
(Ref. Cohort '98)				
Female		0.432***	0.108***	0.107***
(Ref. Male)				
Parental education:				
Leaving Certificate		1.078	0.984	1.014
Post-secondary		1.028	0.876	0.899±
Degree		0.793***	0.593***	0.671***
(Ref. Lower secondary)				
Social class:				
Professional		0.966	0.942	0.810±
Managerial		0.977	0.945	0.758***
Other non-manual		1.086	1.028	0.870
Skilled manual		0.992	0.919	0.896
Non-employed		1.024	0.991	1.030
(Ref. Semi/unskilled manual)				
Experiencing financial strain		1.048	1.028	0.977
Parent(s) born outside Ireland		1.029	0.919	1.015
Lone-parent family		1.084±	1.060	1.050
(Ref. Two-parent family)				
Large family		0.920*	0.922±	0.920*
(Ref. 2 or fewer siblings)				
Disability/illness		1.059	1.035	1.038
Urban location		1.088*	1.072*	1.073*
(Ref. Rural)				
Social/private rented tenure		1.042	1.109	1.114*
(Ref. Own with/without mortgage)				
Female* Cohort '08			15.201***	15.407***
Leaving Certificate* Cohort '08			1.032	
Post-secondary* Cohort '08			1.068	
Degree* Cohort '08			1.331*	
Professional* Cohort '08				1.404*
Managerial* Cohort '08				1.612***
Non-manual* Cohort '08				1.415**
Skilled* Cohort '08				1.040
Non-employed* Cohort '08				0.892
Strain* Cohort '08				1.261*
Log likelihood		-14473.623	-10043.819	-10028.009
N		12,	717	

 Source:
 Growing Up in Ireland Cohorts '98 and '08.

 Note:
 *** p<.001; ** p<.01; * p<.05; + p<.10.</td>

TABLE 3.9 RELATIONSHIP BETWEEN SCREEN TIME (TV AND OTHER) AND OTHER DAY-TO-DAY ACTIVITIES

	Organised sports (Odds ratio)	Low levels of hard exercise (Odds ratio)	Low levels of light exercise (Odds ratio)	Cultural activities (Odds ratio)	Organised clubs (Odds ratio)
Cohort '08	1.287***	0.190***	0.181***	1.318***	0.708***
(Ref. Cohort '98)					
Time watching TV:					
1–2 hours	1.056	1.260***	1.132*	0.892*	1.028
2–3 hours	0.864*	1.395***	1.170**	0.780***	1.084
3–4 hours	0.888±	1.591***	1.145±	0.757***	0.995
4+ hours	0.675***	1.682***	1.555***	0.702***	0.825*
(Ref. <1 hour)					
Other screen time:					
<1 hour	0.998	0.960	0.921	0.920	1.013
1–2 hours	0.944	1.278***	1.051	0.757***	0.786***
3+ hours	0.843**	1.357***	1.060	0.727***	0.819**
(Ref. None)					
Nagelkerke R ²	0.140	0.258	0.234	0.166	0.020

Source: Growing Up in Ireland Cohorts '98 and '08.

Note: *** p<.001; ** p<.01; * p<.05; + p<.10.

3.4 CONCLUSIONS

This chapter has looked at changes in day-to-day activities among 13-year-olds. There is an increase over time in the proportion who attend organised sports at least weekly as well as a reduction in the numbers with low levels (two or fewer days a fortnight) of hard and light exercise. There is some narrowing of the gender gap in sports and hard exercise, though differences remain substantial. Differences by social background and migrant origin in organised sports and hard exercise levels are substantial, but the differences by social background decline somewhat over time.

There is evidence of a stable, if not increasing, level of participation in cultural activities (such as drama and dance), though gender and social background differences remain large. There is a slight decline in the numbers taking part in organised groups such as youth clubs and Scouts/Guides, which may be, at least partly, explained by the persistence of pandemic restrictions at the time of the survey. Not surprisingly, there is a significant shift away from traditional media such as television towards other screen-based activities, with a swing in the gender pattern towards longer screen time among girls. Further research is needed to unpack changes in the types of screen use across different groups of young people. There is some evidence of less involvement in sports, exercise and cultural activities among those with greater screen time.

TABLE A3.1SENSITIVITY ANALYSES OF SPORTS AND CULTURAL ACTIVITIES TO INCLUDE
HOUSEHOLD INCOME QUINTILE (ODDS RATIOS)

	Organised sports (weekly)	Low levels of hard exercise	Low levels of light exercise	Cultural activities (weekly)	Organised groups (weekly)
Cohort '08	1.383***	0.303***	0.147***	2.324**	0.731***
(Ref. Cohort '98)					
Household income quintile:					
Quintile 2	1.174*	1.126	0.958	1.008	1.010
Quintile 3	1.392***	0.975	0.885	1.271*	1.091
Quintile 4	1.668***	0.778**	0.839*	1.452***	1.013
Quintile 5	2.343***	0.541***	0.714***	2.116***	0.861
Income missing	1.198±	0.727*	0.754***	1.568***	0.950
(Ref. Lowest quintile)					
Quintile 2* Cohort '08	1.103	1.025	1.102	0.912	0.972
Quintile 3* Cohort '08	1.389*	0.722±	1.045	0.844	1.008
Quintile 4* Cohort '08	1.075	1.105	1.472*	0.761±	0.796
Quintile 5* Cohort '08	0.990	1.056	1.708***	0.587***	1.185
Income missing* Cohort '08	1.053	1.509±	1.399±	0.723±	0.924
Nagelkerke R ²	0.116	0.146	0.162	0.147	0.015
Ν	13,276	13,221	13,220	13,509	13,235

Source: Growing Up in Ireland Cohorts '98 and '08.

Note: *** p<.001; ** p<.01; * p<.05; + p<.10. The models also control for gender, family structure, family size, migrant status, urban/rural location and housing tenure.

TABLE A3.2SENSITIVITY ANALYSES OF SCREEN TIME TO INCLUDE HOUSEHOLD INCOME
QUINTILE (ODDS RATIOS)

	Television	Games	Other screen time
Cohort '08	0.294***	0.562***	0.930
(Ref. Cohort '98)			
Household income quintile:			
Quintile 2	0.931	0.850*	1.141±
Quintile 3	0.853*	0.967	0.872±
Quintile 4	0.753***	0.795***	0.753***
Quintile 5	0.581***	0.582***	0.601***
Income missing	0.818*	0.827*	0.888
(Ref. Lowest quintile)			
Quintile 2* Cohort '08	0.833±	1.253*	0.898
Quintile 3* Cohort '08	0.822±	0.982	1.296*
Quintile 4* Cohort '08	0.958	1.177	1.416**
Quintile 5* Cohort '08	1.135	1.257*	1.695***
Income missing* Cohort '08	0.805	1.175	1.261±
Log likelihood	-15087.632	-14559.617	-13635.387
Ν	12,700	12,726	12,717

Source: Growing Up in Ireland Cohorts '98 and '08.

Note: *** p<.001; ** p<.01; * p<.05; + p<.10. The models also control for gender, family structure, family size, migrant status, urban/rural location and housing tenure.

CHAPTER 4

Educational experiences

4.1 INTRODUCTION

As discussed in Chapter 1, the educational experiences of 13-year-olds from Cohort '08 have been impacted by two sets of factors: junior cycle reform which has affected the number and content of subjects studied as well as the assessment approaches experienced; and the disruption to learning caused by the pandemic, a time when most of this cohort were making the transition to second-level education. Changes in teaching and learning at primary level will also have influenced student experiences of a subject area. The vast majority of both cohorts were in first or second year of second-level education; no further detail is available on the 'other' category but this is likely to include those in special schools (which are counted as primary level for official purposes) and those who were being homeschooled. This chapter looks at changes over time in school engagement, that is, whether young people like school, and subject engagement, the extent to which young people find English, Irish, Maths and Science interesting. Young people were asked about these subjects as the vast majority take them at junior cycle level. As in Chapters 2 and 3, multivariate models are used to assess the extent of changes over time and whether patterns by gender and social background have altered.

4.2 SCHOOL ENGAGEMENT

Figure 4.1 shows some change over time in adolescent attitudes to school, with a reduction in those saying they like school very much (from 29% to 21%) and an accompanying increase in liking school 'quite a bit' or a 'bit'. The numbers who dislike or even hate school remain stable over time.


FIGURE 4.1 SCHOOL ENGAGEMENT BY COHORT (ACTUAL PERCENTAGES)

Source: GUI Cohort '98 and '08.

Note: C98: Cohort '98; C08: Cohort '08.

Table 4.1 shows that this shift in attitudes over time is not related to the changing composition of young people and their families (Model 2). Girls have more positive attitudes to school than boys. However, this pattern changes over time (Model 3), with a significant narrowing of the gender gap in school attitudes among Cohort '08.¹¹ To explore whether this trend reflects changes in wellbeing, a simple analysis of the relationship between emotional difficulties (the SDQ subscale) and attitudes to school was carried out (Table A4.2).¹² The gender gap in emotional difficulties widened over the decade concerned, from 0.38 points to 0.64 points. For Cohort '98, the raw gender gap in school engagement was 0.241, with this gap widening slightly to 0.281 when emotional difficulties and year group are taken into account. For Cohort '08, however, the raw gender gap is much smaller at 0.075, with this gap increasing to 0.150 when emotional difficulties and year group are included in the model. Thus, emotional difficulties negatively impact on girls' school engagement in both cohorts but appear to do so to a much greater extent among the younger cohort.

School engagement is highly structured by all aspects of family background, with more positive attitudes among those with graduate parents as well as those from professional/managerial and higher-income households. The social gradient in school engagement is broadly stable over time, except for a slight narrowing of the gap between the top income quintile and the remainder (Table A4.1). Being under financial strain is associated with more negative attitudes to school, even taking

¹¹ Descriptive analyses show that 24 per cent of girls like school 'very much' compared with 22 per cent of boys. For Cohort '98, the figures were 35 per cent and 23 per cent respectively.

¹² For ease of comparison across models, an OLS regression model was used.

account of parental education and social class. Attitudes are also less positive among those from lone-parent families, those living in rented accommodation and urban dwellers. Migrant-origin adolescents do not differ from their peers in their attitudes to school. The strongest relationship is found between disability and school attitudes, with particularly negative attitudes to school among those with a disability.

TABLE 4.1GENERALISED LINEAR MODELS (CUMULATIVE LOGIT FUNCTION) OF POSITIVE
ATTITUDES TO SCHOOL (ODDS RATIOS)

	Raw differences (1)	Differences controlling for family and child factors (2)	Changes in the effect of gender and education (3)	Changes in the effect of class and strain (4)
Cohort '08	0.807***	0.795***	1.098	1.002
(Ref. Cohort '98)				
Female		1.452***	1.750***	1.745***
(Ref. Male)				
Parental education:				
Leaving Certificate		1.195**	1.237**	1.211**
Post-secondary		1.254***	1.323***	1.263***
Degree		1.377***	1.531***	1.405***
(Ref. Lower secondary)				
Social class:				
Professional		1.504***	1.515***	1.609***
Managerial		1.336***	1.346***	1.413***
Other non-manual		1.072	1.088	1.065
Skilled manual		1.088	1.104	1.069
Non-employed		1.111	1.115	1.124
(Ref. Semi/unskilled manual)				
Experiencing financial strain		0.913*	0.912*	0.902±
Parent(s) born outside Ireland		1.100	1.106±	1.100
Lone-parent family		0.818***	0.821***	0.822***
(Ref. Two-parent family)				
Large family		0.942	0.940	0.940
(Ref. 2 or fewer siblings)				
Disability/illness		0.661***	0.662***	0.663***
Urban location		0.802***	0.806***	0.804***
(Ref. Rural)				
Social/private rented tenure		0.898*	0.891*	0.890*
(Ref. Own with/without mortgage)				
Female* Cohort '08			0.664***	0.662***
Leaving Certificate* Cohort '08			0.955	
Post-secondary* Cohort '08			0.889	
Degree* Cohort '08			0.802±	
Professional* Cohort '08				0.873
Managerial* Cohort '08				0.883
Non-manual* Cohort '08				1.033
Skilled* Cohort '08				1.078
Non-employed* Cohort '08				0.985
Strain* Cohort '08				1.038
Log likelihood		-12090 570	-8640 801	-8640 226
N		12 1	563	50.0.220

 Source:
 Growing Up in Ireland Cohorts '98 and '08.

 Note:
 *** p<.001; ** p<.01; * p<.05; + p<.10.</td>

4.3 SUBJECT ENGAGEMENT

At both timepoints, 13-year-olds express higher levels of interest in Science and English with lower levels of interest shown in Irish (Figure 4.2). Members of Cohort '08, who had experienced junior cycle reform, reported higher levels of interest in Maths, English and Science than those from Cohort '98. At the same time, there was also a small increase for Maths and English in those finding the subjects 'not interesting' as opposed to 'OK'. Attitudes to Irish differed in showing a sizeable increase in the percentage finding the subject 'not interesting' (from 32% to 42%).



FIGURE 4.2 PERCEIVED INTEREST IN MATHS, ENGLISH, IRISH AND SCIENCE BY COHORT (ACTUAL PERCENTAGES)

Source: GUI Cohort '98 and '08.

4.3.1 Maths engagement

Table 4.2 shows that even taking account of the shift in adolescent profile, there was an increase in interest in Maths and a slight increase in those finding the subject not interesting compared to those who find it 'OK'. Girls are less likely than boys to find Maths interesting and more likely to describe it as not interesting. Furthermore, the gender gap in lack of interest increases between Cohort '98 and Cohort '08 (Model 3), with the predicted probability increasing from 21 to 26 per cent among girls and remaining stable at 19 per cent for boys. Maths engagement increases with levels of parental education with this pattern remaining fairly stable over time. There is no clear gradient by social class when parental education is taken into account and only those from the highest-income group differ from their peers in attitudes to Maths. Migrant-origin young people are equally likely to find Maths interesting but less likely to describe it as not interesting rather than OK. Levels of interest in Maths are lower among those from lone-parent families and, especially, among those with a disability. Urban adolescents tend to be more polarised in their attitudes to Maths than their rural peers.

TABLE 4.2 MULTINOMIAL LOGIT MODELS OF ATTITUDE TO MATHS (ODDS RATIOS) (BASE CATEGORY: OK)

	Raw differences (1)		Differences controlling for family and child factors (2)		Changes in the effect of gender and education (3)		Changes in the effect of class and strain (4)	
	Interesting	Not interesting	Interesting	Not interesting	Interesting	Not interesting	Interesting	Not interesting
Cohort '08	1.807***	1.576***	1.800***	1.571***	1.903***	1.746***	2.059***	1.134
(Ref. Cohort '98)								
Female (Ref. Male)			0.809***	1.162**	0.838***	1.064	0.837***	1.056
Parental education:								
Leaving Certificate			1.178*	1.075	1.134	1.129	1.173*	1.062
Post-secondary			1.241**	1.044	1.147	1.059	1.233*	1.034
Degree			1.452***	1.089	1.589***	1.242	1.460***	1.085
(Ref. Lower secondary)								
Social class: Professional			0.894	0.693**	0.912	0.703**	0.965	0.564***
Managerial			0.905	0.886	0.923	0.896	0.979	0.829±
Other non-manual			0.903	0.810*	0.915	0.814*	0.999	0.742*
Skilled manual			0.819*	0.954	0.826*	0.953	0.810±	0.830
Non-employed			0.925	0.773*	0.929	0.772*	0.885	0.631***
(Ref. Semi/unskilled manual)								
Experiencing financial strain			0.947	1.008	0.951	1.015	0.992	1.000
Parent(s) born outside Ireland			0.998	0.718***	1.000	0.714***	0.986	0.711***
Lone-parent family			1.111±	1.506***	1.114±	1.507***	1.121±	1.507***
(Ref. Two-parent family)								
Large family			1.089±	0.950	1.087±	0.950	1.089±	0.948
(Ref. 2 or fewer siblings)								
Disability/illness			0.824***	1.220***	0.823***	1.216***	0.823***	1.221***

Source: Growing Up in Ireland Cohorts '98 and '08.

TABLE 4.2(CONTINUED)

	Raw dif	ferences (1)	Differences con and child	trolling for family factors (2)	Changes in the e educa	ffect of gender and ation (3)	Changes in the e stra	effect of class and in (4)
	Interesting	Not interesting	Interesting	Not interesting	Interesting	Not interesting	Interesting	Not interesting
Urban location			1.172***	1.267***	1.173***	1.268***	1.175***	1.263
(Ref. Rural)								
Social/private rented tenure			0.826	0.942	0.975	0.942	0.980	0.946
(Ref. Own with/without mortgage)								
Female* Cohort '08					0.943	1.226*	0.941	1.227*
Leaving Certificate* Cohort '08					1.158	0.866		
Post-secondary* Cohort '08					1.108	0.872		
Degree* Cohort '08					0.796	0.682*		
Professional* Cohort '08							0.862	1.476±
Managerial* Cohort '08							0.842	1.134
Non-manual* Cohort '08							0.813	1.186
Skilled* Cohort '08							1.064	1.374±
Non-employed* Cohort '08							1.174	1.635±
Strain* Cohort '08							0.872	0.777
Nagelkerke R ²	0.020		0.044		0.046		0.046	
Ν				13,	098			

Source: Growing Up in Ireland Cohorts '98 and '08.

4.3.2 Engagement in English

In keeping with the descriptive findings, attitudes to English become a little more polarised over time and this is not accounted for by changes in the profile of young people (Table 4.3). Girls are much more positive about English as a subject than boys and the gender gap remains stable over time. Attitudes are more polarised among those whose parents have higher levels of education and there is some evidence that the gap between those whose parents have post-secondary or tertiary qualifications and others becomes smaller over time. Those who have a disability or are from a large family are less interested in English while those in rental accommodation are slightly more interested than might be expected given their other characteristics. Migrant-origin youth are more likely to characterise English as OK rather than interesting or not interesting.

TABLE 4.3 MULTINOMIAL LOGIT MODELS OF ATTITUDE TO ENGLISH (ODDS RATIOS) (BASE CATEGORY: OK)

	Raw differences (1)		Differences con and child	Differences controlling for family and child factors (2)		fect of gender and tion (3)	Changes in the effect of class and strain (4)	
	Interesting	Not interesting	Interesting	Not interesting	Interesting	Not interesting	Interesting	Not interesting
Cohort '08	1.564***	1.737***	1.543***	1.701***	2.195***	1.705*	1.858***	1.532*
(Ref. Cohort '98)								
Female (Ref. Male)			1.350***	0.857**	1.357***	0.915	1.350***	0.915
Parental education:								
Leaving Certificate			1.183*	1.488***	1.262**	1.504**	1.179*	1.497***
Post-secondary			1.188*	1.398**	1.285**	1.254	1.161*	1.412**
Degree			1.361***	1.472**	1.670***	1.426*	1.343***	1.473**
(Ref. Lower secondary)								
Social class: Professional			0.874	0.833	0.891	0.836	1.086	0.787
Managerial			0.895	0.910	0.910	0.913	0.999	0.807
Other non-manual			0.883±	0.836	0.897	0.838	0.843±	0.724*
Skilled manual			0.866±	0.912	0.874±	0.914	1.008	0.868
Non-employed			0.885	0.681**	0.887	0.684**	0.919	0.623**
(Ref. Semi/unskilled manual)								
Experiencing financial strain			0.949	0.955	0.958	0.953	0.979	0.988
Parent(s) born outside Ireland			0.887±	0.773*	0.882±	0.775*	0.892±	0.783*
Lone-parent family			1.072	1.148	1.075	1.151	1.077	1.148
(Ref. Two-parent family)								
Large family			0.868**	0.889	0.867**	0.888	0.864**	0.886±
(Ref. 2 or fewer siblings)								
Disability/illness			0.868**	1.043	0.865**	1.044	0.864**	1.048

Source: Growing Up in Ireland Cohorts '98 and '08.

TABLE 4.3(CONTINUED)

	Raw dif	ferences (1)	Differences con and child	trolling for family I factors (2)	Changes in the e educa	ffect of gender and ation (3)	Changes in the e stra	effect of class and in (4)
	Interesting	Not interesting	Interesting	Not interesting	Interesting	Not interesting	Interesting	Not interesting
Urban location			1.021	1.005	1.025	1.006	1.020	1.003
(Ref. Rural)								
Social/private rented tenure			1.144*	0.980	0.838	0.983	1.143*	0.986
(Ref. Own with/without mortgage)								
Female* Cohort '08					1.002	0.870	0.999	0.867
Leaving Certificate* Cohort '08					0.838	0.895		
Post-secondary* Cohort '08					0.732*	1.199		
Degree* Cohort '08					0.562***	1.031		
Professional* Cohort '08							0.645**	1.102
Managerial* Cohort '08							0.786±	1.258
Non-manual* Cohort '08							1.145	1.404
Skilled* Cohort '08							0.700±	1.074
Non-employed* Cohort '08							0.931	1.199
Strain* Cohort '08							0.883	0.859
Nagelkerke R ²	0.015		0.032		0.035		0.034	
Ν				13,	087			

Source: Growing Up in Ireland Cohorts '98 and '08.

4.3.3 Engagement in Irish

The patterns of increased numbers finding Irish not interesting holds even when the changing profile of adolescents and their families is taken into account (Model 2, Table 4.4). Girls have higher levels of interest in the subject than boys, though the gender gap in finding Irish interesting narrows over time. As with English, attitudes to Irish are more polarised among those from more highly educated families. Attitudes do not vary markedly by income but are less positive among the lowest-income group and most positive among the highest-income quintile. There is some evidence of a stronger social class gradient in attitudes among Cohort '08 than Cohort '98 (Model 4). Migrant-origin young people are less likely than their peers to consider Irish interesting. Lack of interest in Irish is more prevalent among those with a disability, those from lone-parent families and those in rented accommodation.

TABLE 4.4 MULTINOMIAL LOGIT MODELS OF ATTITUDE TO IRISH (ODDS RATIOS) (BASE CATEGORY: OK)

	Raw differences (1)		Differences con and child	Differences controlling for family and child factors (2)		Changes in the effect of gender and education (3)		Changes in the effect of class and strain (4)	
	Interesting	Not interesting	Interesting	Not interesting	Interesting	Not interesting	Interesting	Not interesting	
Cohort '08	1.350***	1.675***	1.321***	1.582***	1.292	1.781***	1.069	1.599***	
(Ref. Cohort '98)									
Female (Ref. Male)			1.251***	0.836***	1.401***	0.830***	1.385***	0.827***	
Parental education:									
Leaving Certificate			1.268*	1.256**	1.139	1.270**	1.290**	1.258**	
Post-secondary			1.272**	1.202*	1.242±	1.162	1.290**	1.182*	
Degree			1.396***	1.371***	1.402**	1.548***	1.424***	1.355***	
(Ref. Lower secondary)									
Social class: Professional			0.841	0.826±	0.847	0.839±	0.641**	0.935	
Managerial			0.910	0.833*	0.917	0.847*	0.830±	0.865	
Other non-manual			0.932	0.843*	0.944	0.851±	0.732**	0.740**	
Skilled manual			0.771**	0.850±	0.782*	0.855±	0.647***	0.925	
Non-employed			0.868	1.000	0.865	1.007	0.767±	1.025	
(Ref. Semi/unskilled manual)									
Experiencing financial strain			0.998	0.965	1.000	1.002	1.042	0.996	
Parent(s) born outside Ireland			0.570***	1.045	0.575***	1.043	0.577***	1.053	
Lone-parent family			0.987	1.312***	0.984	1.314***	0.982	1.311***	
(Ref. Two-parent family)									
Large family			1.007	0.987	1.002	0.986	1.001	0.982	
(Ref. 2 or fewer siblings)									
Disability/illness			1.053	1.363***	1.059	1.359***	1.064	1.363***	

Source: Growing Up in Ireland Cohorts '98 and '08.

TABLE 4.4(CONTINUED)

	Raw dif	ferences (1)	Differences con and child	trolling for family I factors (2)	Changes in the e educ	ffect of gender and ation (3)	Changes in the e stra	effect of class and in (4)
	Interesting	Not interesting	Interesting	Not interesting	Interesting	Not interesting	Interesting	Not interesting
Urban location			0.941	1.083±	0.940	1.084±	0.936	1.080±
(Ref. Rural)								
Social/private rented tenure			1.150±	1.348***	1.132±	1.336***	1.163*	1.354***
(Ref. Own with/without mortgage)								
Female* Cohort '08					0.765***	1.004	0.775**	1.004
Leaving Certificate* Cohort '08					1.677*	1.042		
Post-secondary* Cohort '08					1.154	0.995		
Degree* Cohort '08					1.069	0.739±		
Professional* Cohort '08							1.805**	0.825
Managerial* Cohort '08							1.279	0.947
Non-manual* Cohort '08							1.930***	1.417*
Skilled* Cohort '08							1.587*	0.846
Non-employed* Cohort '08							1.382	0.962
Strain* Cohort '08							0.783	0.964
Nagelkerke R ²	0.014		0.044		0.047		0.048	
Ν				12,3	131			

Source: Growing Up in Ireland Cohorts '98 and '08.

4.3.4 Science engagement

Figure 4.2 had shown a significant increase in the level of interest expressed in Science over time, a pattern that is not explained by the changing composition of adolescents (Model 2, Table 4.5). Girls express lower levels of interest in Science than boys and this gender gap widens somewhat over time. Expressing this in terms of predicted probabilities, 61 per cent of boys and 58 per cent of girls in Cohort '98 found Science interesting; by Cohort '08, the figures were 71 per cent and 65 per cent respectively. Science engagement is structured by social background, being higher for more educated and higher-income families, and these differences are stable over time. Migrant-origin young people express more interest in Science while those experiencing financial strain, from a larger family or having a disability have lower levels of interest. Levels of interest are slightly higher among those in rented accommodation than might be expected given their profile in terms of other measures of family background.

TABLE 4.5 MULTINOMIAL LOGIT MODELS OF ATTITUDE TO SCIENCE (ODDS RATIOS) (BASE CATEGORY: OK)

	Raw differences (1)		Differences cont and child	Differences controlling for family and child factors (2)		Changes in the effect of gender and education (3)		Changes in the effect of class and strain (4)	
	Interesting	Not interesting	Interesting	Not interesting	Interesting	Not interesting	Interesting	Not interesting	
Cohort '08	1.497***	1.105	1.353***	1.104	1.503**	1.264	1.364*	1.157	
(Ref. Cohort '98)									
Female (Ref. Male)			0.875***	1.093	0.981	1.148±	0.979	1.141	
Parental education:									
Leaving Certificate			1.100	0.823±	1.050	0.776*	1.103	0.826±	
Post-secondary			1.433***	0.837	1.451***	0.965	1.447***	0.833±	
Degree			1.572***	0.802±	1.615***	0.840	1.583***	0.798±	
(Ref. Lower secondary)									
Social class: Professional			1.043	0.747±	1.053	0.748±	1.015	0.846	
Managerial			1.127	0.952	1.138	0.953	1.064	0.937	
Other non-manual			1.108	1.042	1.125	1.056	1.023	0.967	
Skilled manual			0.890	0.961	0.903	0.971	0.889	0.981	
Non-employed			1.075	0.889	1.080	0.885	0.961	0.856	
(Ref. Semi/unskilled manual)									
Experiencing financial strain			0.861*	0.872	0.860*	0.873	0.863*	0.909	
Parent(s) born outside Ireland			1.224*	1.145	1.231*	1.144	1.237*	1.158	
Lone-parent family			0.987	0.910	0.988	0.908	0.991	0.912	
(Ref. Two-parent family)									
Large family			0.759***	0.920	0.758***	0.919	0.755***	0.916	
(Ref. 2 or fewer siblings)									
Disability/illness			0.803***	0.940	0.806***	0.943	0.806***	0.943	

Source: Growing Up in Ireland Cohorts '98 and '08.

TABLE 4.5(CONTINUED)

	Raw dif	ferences (1)	Differences con and child	trolling for family I factors (2)	Changes in the e educ	ffect of gender and ation (3)	Changes in the e	effect of class and in (4)
	Interesting	Not interesting	Interesting	Not interesting	Interesting	Not interesting	Interesting	Not interesting
Urban location			1.060	1.286***	1.061	1.287***	1.061	1.285***
(Ref. Rural)								
Social/private rented tenure			1.196**	1.065	1.179*	1.051	1.193**	1.065
(Ref. Own with/without mortgage)								
Female* Cohort '08					0.755***	0.881	0.757**	0.883
Leaving Certificate* Cohort '08					1.300	1.146		
Post-secondary* Cohort '08					0.997	0.743		
Degree* Cohort '08					0.963	0.885		
Professional* Cohort '08							1.084	0.780
Managerial* Cohort '08							1.160	1.055
Non-manual* Cohort '08							1.248	1.236
Skilled* Cohort '08							1.014	0.951
Non-employed* Cohort '08							1.349	1.130
Strain* Cohort '08							0.973	0.831
Nagelkerke R ²	0.009		0.035		0.037		0.037	
Ν				12,3	395			

Source: Growing Up in Ireland Cohorts '98 and '08.

4.4 CONCLUSIONS

This chapter has explored changes over time in young people's engagement with school in general and with particular subjects. Such changes must be seen against the backdrop of junior cycle reform as well as the disruption to education caused by the pandemic. While attitudes to school are broadly stable among boys, girls become somewhat less positive about school in general, apparently because of the impact of greater emotional difficulties. Further research is needed to explore whether such difficulties reflect the impact of the pandemic and/or other aspects of social change. The findings highlight persistent differences in school engagement by socio-economic background, with parental education, social class, income, experience of financial strain and family structure all significantly related to attitudes to school. Levels of school engagement are found to be much lower among young people with a disability.

Junior cycle reform has meant a change in the content of subjects and how they are assessed. The findings point to increased interest in Science, English and Maths (but not Irish). At the same time, there has been a slight increase in the numbers finding English and Maths not interesting as opposed to OK. There are significant gender differences in subject engagement, with girls more positive about English and Irish and boys more positive about Maths and Science. Furthermore, there is concerning evidence of a slight widening of the gender gap in STEM interest over time, echoing the widening of the gender gap in Maths attitudes among this cohort at the age of nine (Smyth, 2022). Engagement across all four subjects analysed tends to be significantly higher among those with more highly educated parents, though other aspects of family background are not as clearly related to subject attitudes. In keeping with their attitudes to school overall, young people with a disability tend to be significantly more negative about English, Irish, Science and Maths.

TABLE A4.1SENSITIVITY ANALYSES OF ATTITUDE TO SCHOOL TO INCLUDE HOUSEHOLD INCOME
QUINTILE (ODDS RATIOS)

	Attitude to school
Cohort '08	1.183*
(Ref. Cohort '98)	
Household income quintile:	
Quintile 2	1.242**
Quintile 3	1.198*
Quintile 4	1.349***
Quintile 5	1.739***
Income missing	1.661***
(Ref. Lowest quintile)	
Quintile 2* Cohort '08	0.862
Quintile 3* Cohort '08	0.897
Quintile 4* Cohort '08	0.873
Quintile 5* Cohort '08	0.732**
Income missing* Cohort '08	0.701**
Log likelihood	-12057.005
Ν	12,563

 Source:
 Growing Up in Ireland Cohorts '98 and '08.

 Note:
 *** p<.001; ** p<.01; * p<.05; + p<.10. Th</td>

*** p<.001; ** p<.01; * p<.05; + p<.10. The models also control for gender, family structure, family size, migrant status, urban/rural location and housing tenure.

TABLE A4.2 OLS REGRESSION MODELS OF RELATIONSHIP BETWEEN SOCIO-EMOTIONAL WELLBEING AND ATTITUDES TO SCHOOL VELLBEING VELBEING VELLBEING

	Coho	rt '98	Cohort '08		
	Model 1	Model 2	Model 1	Model 2	
Constant	3.635	3.872	3.607	3.927	
Female	0.241***	0.281***	0.072**	0.150***	
In second year of second-level education		-0.277***		-0.178***	
SDQ emotional problems		-0.058***		-0.101***	
Adjusted R ²	0.013	0.042	0.001	0.062	
Ν	7	.374	6,265		

Source: Growing Up in Ireland Cohorts '98 and '08.

TABLE A4.3SENSITIVITY ANALYSES OF ATTITUDE TO MATHS TO INCLUDE HOUSEHOLD INCOME
QUINTILE (ODDS RATIOS)

	Interesting	Not interesting
Cohort '08	2.186***	1.668***
(Ref. Cohort '98)		
Household income quintile:		
Quintile 2	1.005	1.252*
Quintile 3	0.880	0.986
Quintile 4	1.101	1.114
Quintile 5	1.205*	0.830
Income missing	1.090	0.895
(Ref. Lowest quintile)		
Quintile 2* Cohort '08	0.745*	0.598***
Quintile 3* Cohort '08	0.927	0.948
Quintile 4* Cohort '08	0.988	0.861
Quintile 5* Cohort '08	0.847	0.817
Income missing* Cohort '08	0.756±	0.932
Nagelkerke R ²	0.0	43
Ν	13,0	098

Source: Growing Up in Ireland Cohorts '98 and '08.

Note: *** p<.001; ** p<.01; * p<.05; + p<.10. The models also control for gender, family structure, family size, migrant status, urban/rural location and housing tenure.

TABLE A4.4SENSITIVITY ANALYSES OF ATTITUDE TO ENGLISH TO INCLUDE HOUSEHOLD INCOME
QUINTILE (ODDS RATIOS)

	Interesting	Not interesting
Cohort '08	1.787***	1.450*
(Ref. Cohort '98)		
Household income quintile:		
Quintile 2	1.246**	0.935
Quintile 3	1.132	0.883
Quintile 4	1.365***	1.376*
Quintile 5	1.289**	1.029
Income missing	1.290*	1.033
(Ref. Lowest quintile)		
Quintile 2* Cohort '08	0.836	1.337
Quintile 3* Cohort '08	0.952	1.720**
Quintile 4* Cohort '08	0.792±	0.959
Quintile 5* Cohort '08	0.865	1.507*
Income missing* Cohort '08	0.843	1.204
Nagelkerke R ²	0.031	
Ν	13,087	

Source: Growing Up in Ireland Cohorts '98 and '08.

Note: *** p<.001; ** p<.01; * p<.05; + p<.10. The models also control for gender, family structure, family size, migrant status, urban/rural location and housing tenure.

TABLE A4.5SENSITIVITY ANALYSES OF ATTITUDE TO IRISH TO INCLUDE HOUSEHOLD INCOME
QUINTILE (ODDS RATIOS)

	Interesting	Not interesting
Cohort '08	1.709***	1.697***
(Ref. Cohort '98)		
Household income quintile:		
Quintile 2	1.495***	1.271**
Quintile 3	1.398**	1.234**
Quintile 4	1.462***	1.109
Quintile 5	1.650***	1.302***
Income missing	1.198	1.385**
(Ref. Lowest quintile)		
Quintile 2* Cohort '08	0.764	0.791±
Quintile 3* Cohort '08	1.116	1.013
Quintile 4* Cohort '08	0.943	0.982
Quintile 5* Cohort '08	0.758	0.940
Income missing* Cohort '08	0.960	0.806
Nagelkerke R ²	0.046	
Ν	12,131	

Source: Growing Up in Ireland Cohorts '98 and '08.

Note: *** p<.001; ** p<.01; * p<.05; + p<.10. The models also control for gender, family structure, family size, migrant status, urban/rural location and housing tenure.

TABLE A4.6SENSITIVITY ANALYSES OF ATTITUDE TO SCIENCE TO INCLUDE HOUSEHOLD INCOME
QUINTILE (ODDS RATIOS)

	Interesting	Not interesting
Cohort '08	2.100***	1.390*
(Ref. Cohort '98)		
Household income quintile:		
Quintile 2	1.241*	1.119
Quintile 3	1.421***	0.865
Quintile 4	1.530***	1.352*
Quintile 5	1.687***	1.043
Income missing	1.555***	1.087
(Ref. Lowest quintile)		
Quintile 2* Cohort '08	0.795	0.742
Quintile 3* Cohort '08	0.815	1.309
Quintile 4* Cohort '08	0.793	0.719
Quintile 5* Cohort '08	0.829	0.927
Income missing* Cohort '08	0.577**	0.423**
Nagelkerke R ²	0.030	
Ν	12,395	

Source: Growing Up in Ireland Cohorts '98 and '08.

Note: *** p<.001; ** p<.01; * p<.05; + p<.10. The models also control for gender, family structure, family size, migrant status, urban/rural location and housing tenure.

CHAPTER 5

Conclusions

5.1 BACKGROUND TO THE STUDY

This study takes advantage of the two-cohort nature of the GUI study to explore changes in the lives of adolescents over the period 2011/12–2021/22, building on an earlier study on changes in the lives of nine-year-olds (Smyth, 2022). The report focuses on the key domains of social relationships, day-to-day activities and educational experiences. The families of young people from Cohort '08 are better educated, more culturally diverse, smaller and less likely to be under financial strain than their counterparts from Cohort '98. All of these factors would be expected to influence adolescent outcomes so the analyses look at whether changes in outcomes are evident over and above these compositional shifts. As well as changes in the profile of families, significant social and policy changes took place over this decade, including reform at junior cycle, increasing digitalisation of daily lives and the disruption caused by the pandemic to all aspects of the lives of adolescents and their families.

The main research questions addressed by the study are:

- How have the quality of relationships, experience of learning and activities engaged in by adolescents changed over the course of a decade (2011/12– 2021/22)?
- 2. To what extent do any such changes reflect differences in the family characteristics of the young people?
- 3. Are any such changes more evident for boys or girls or for young people from different social backgrounds? Is differentiation by gender and social background in adolescents' social worlds less evident for the younger cohort than previously?

This chapter outlines the main findings emerging from the study (summarised in Table 5.1) and discusses the implications of these findings for policy development.

5.2 CHANGES IN SOCIAL RELATIONSHIPS

International research has highlighted a shift over time in the ideology of parenting, with a more child-centred approach, less emphasis on obedience and more quality time spent by parents with children (Sayer et al., 2004; Altintas and Sullivan, 2017; Ryan et al., 2020; Chambers and Gracia, 2021). A comparison of the two GUI cohorts suggests improved parent-child relations in Ireland, with a growing democratisation of family life. Over a decade, mothers and fathers reported improved levels of

closeness with their nine-year-old children (Smyth, 2022). Although levels of conflict were broadly stable over time in middle childhood, analyses at the age of 13 indicate sizeable reductions in the extent of parent-child conflict among both mothers and fathers and an improvement in how responsive young people felt their mother was to their needs. During the pandemic restrictions, the majority of parents and young people in the cohort reported enjoying the additional time they spent with their family members (GUI Study Team, 2021). Further research could usefully explore whether this closeness played any role in reduced parent-child conflict. Conflict levels reduced across all social groups, with the greatest improvement among more disadvantaged groups. The exception was those families experiencing financial strain where conflict levels were stable over time. Conflict reduced more for boys than for girls; preliminary analysis suggests this may be, at least partly, related to a gendered increase in emotional difficulties but further research is needed to examine whether this reflects the impact of the pandemic or other factors. The parental approach to dealing with behaviour has also changed over time, with less emphasis on shouting at or grounding young people and more focus on explaining what they did wrong.

Peers adopt a more important role as young people move into adolescence (Brown and Larson, 2009), though the transition to second-level education can disrupt existing friendship networks (Smyth et al., 2004). The pandemic-related restrictions resulted in a very severe disruption to face-to-face contact with peers, with almost half of this cohort meeting their friends less than previously even when schools had re-opened (GUI Study Team, 2021). At the age of nine, members of Cohort '08 reported having somewhat larger friendship groups than those of Cohort '98. However, by the age of 13, this pattern had reversed, with a marked reduction in the number of friends overall and of close friends. This pattern was largely driven by a reduction in peer group size among more disadvantaged groups, though the reduction was also greater for girls than boys. The shift was accompanied by less age diversity in friendship groups, with the younger cohort less likely to socialise with much older peers. The change over time may reflect the impact of the pandemic, as those who curtailed contact with friends to a greater extent had fewer friends later on. However, further research is needed to explore this issue.

Unfortunately, comparable information is not available on friendship quality or the frequency of contact with friends. Instead, the mother-reported peer difficulties SDQ subscale is used as a proxy, with a significant increase found in the level of difficulties over time, though this pattern is largely driven by the increase in the groups of young people with poorer peer relations, namely, those with a disability, from migrant backgrounds or living in rented accommodation. Quantity and quality are interrelated, with more difficulties interacting with others found among those with very few friends.

TABLE 5.1 SUMMARY OF PATTERNS OF CHANGE IN ADOLESCENT EXPERIENCES AND OUTCOMES BETWEEN COHORTS

Change between cohorts	Changes by gender	Changes by social background
Decreased	No gender gap initially but higher for girls in Cohort '08	Little systematic variation by education or class but conflict declined more for disadvantaged families; higher with financial strain, a difference that increases over time
Decreased	Small gender gap initially but higher for girls in Cohort '08	Little systematic variation by education or class but higher with financial strain; declines over time for all social groups, especially non-employed households
Apparent decrease due to compositional change	Less positive for girls, with gap widening over time	Lower for higher educated initially, with the gap narrowing over time; improvement over time for non-employed households
Increased	Initially higher for girls but gender gap reverses over time	Increased over time for all social groups, with greatest increases for disadvantaged households
Apparent decrease due to compositional change	Slight fall for girls over time	Higher for working-class and lower income families; stable over time
Decreased	Smaller among girls in Cohort '08	Greater decline over time among those from non-employed households
Decreased	Smaller among girls in Cohort '08	Greater decline over time for those from non-employed households
Increased; related to compositional change	Slightly lower among girls; stable over time	Lower among professional/managerial and higher income groups; stable over time
Increased	Lower among girls; no change over time	Differentiation by family background; stable over time, though widening gap for those experiencing financial strain
Decreased	Higher for girls; slight narrowing of gender gap over time	Higher among least advantaged groups; some narrowing of gap over time
Decreased	No gender difference	Higher for lower education groups; some narrowing of gap over time
Increased/stable	Much higher for girls; slight narrowing of gender gap over time	Strong social gradient; stable over time except for slight relative increase for non-employed group
Decreased	No gender difference	Little variation by social background; decrease over time is less for working- class group
Declined	Lower for girls; stable over time	Strong social gradient; lower relative decline for more advantaged groups
Declined	Much lower for girls; slight narrowing of gap over time	Lower for professional and graduate families; stable over time
Increased	Lower for girls in Cohort '98 but gender gap reversed by Cohort '08	Lower for more advantaged groups; slight narrowing of gap over time
	Change between cohortsDecreasedDecreasedDecreasedApparent decrease due to compositional changeIncreasedDecreaseDecrease	Change between cohortsChanges by genderDecreasedNo gender gap initially but higher for girls in Cohort '08DecreasedSmall gender gap initially but higher for girls in Cohort '08Apparent decrease due to compositional changeLess positive for girls, with gap widening over timeIncreasedInitially higher for girls but gender gap reverses over timeApparent decrease due to compositional changeSilght fall for girls over timeDecreasedSmaller among girls in Cohort '08DecreasedSmaller among girls in Cohort '08DecreasedSilght fall for girls over timeDecreasedSlightly lower among girls; stable over timeIncreased; related to compositional changeLower among girls; no changeDecreasedHigher for girls; slight narrowing of gender gap over timeDecreasedMuch higher for girls; slight narrowing of gender gap over timeDecreasedMuch higher for girls; slight narrowing of gender gap over timeDecreasedLower for girls; slight narrowing of gender gap over timeDecreasedLower for girls; stable over timeDecreasedLower for girls; slight narrowing of gender gap over timeDecreasedLower for girls; slight narrowing of gender gap over timeDecreasedLower for girls; slight narrowing of gender gap over timeDeclinedLower for girls in Cohort '98 but gender gap reversed by Cohort '08

TABLE 5.1 CONTINUED

	Change between cohorts	Changes by gender	Changes by social background
Educational experiences			
Attitudes to school	Decline in % liking school very much	Girls more positive in Cohort '98; gender gap narrows because of decline for girls	Strong social gradient; stable over time
Interest in Maths	Increased (but some polarisation)	Lower for girls; widening gap over time	Gradient by parental education; stable over time
Interest in English	Increased (but some polarisation)	Higher for girls; stable over time	More polarised by parental education; some narrowing of gap over time
Interest in Irish	Increased % 'not interesting'	Higher for girls; narrowing of gender gap over time	More polarised by parental education; some narrowing of education gap but widening of social class gap over time
Interest in Science	Increased	Lower for girls; slight widening of gap over time	Social gradient; stable over time

5.3 DAY-TO-DAY ACTIVITIES

The study focuses on a number of different activities among adolescents, including involvement in sport and physical exercise, cultural engagement and screen time. At the age of nine, the cohort was asked about the number of days in which they engaged in sport (but were not asked to separate out organised from unstructured sport). Analyses showed a decline in near-daily sports activity over time and an increase in the social gradient in involvement (Smyth, 2022). At the same time, almost all of the nine-year-old children in both cohorts had at least weekly involvement. At age 13, young people were asked separately about organised sports, with the change in answer categories meaning the analyses distinguish weekly involvement and lower levels. Weekly engagement in organised sports increased over time, with the gender and social background differences in involvement remaining stable (Table 5.1). In addition, low levels of engagement in hard and light exercise (the equivalent of two days or fewer per fortnight) are found to decrease over time. While the scale of this change should be interpreted with caution due to changes in the framing of the questions, the results are consistent with those of Woods et al. (2022) who found an increase in the proportion of children and young people meeting the physical activity guidelines and engaging in weekly sport between 2018 and 2022. Further research could usefully examine whether this positive trend is related to more outdoor activities among some groups during the period of pandemic-related restrictions and/or the effect of broader policy development in relation to sports and exercise.

Weekly involvement in organised groups, such as youth clubs or Scouts/Guides, declined over time but analyses suggest this decline may have been, at least partly, related to the ongoing impact of pandemic restrictions at the time of the survey. There was no significant variation by gender for either cohort or little systematic variation by social background. However, the decline in involvement has been less for those from semi/unskilled manual households than for other groups.

Engagement in structured cultural activities (such as music or dance lessons) had declined over time among nine-year-old children. However, there is evidence of stability, if not increase,¹³ in weekly engagement at the age of 13. The very large gender gap in favour of girls narrowed slightly over time but remained sizeable. Engagement was strongly structured by social background (parental education, social class and income), with little change over time, except for an improvement among the non-employed groups. Among nine-year-olds, there was evidence of a decline in the frequency of reading for pleasure over time. Unfortunately, the question wording on reading at age 13 changed between cohorts so is not directly comparable. Nonetheless, it is interesting to look at the patterning of reading behaviour for both groups. In both cohorts, girls read much more frequently than boys and social class differences are greater among boys than girls. While the categories are not comparable, there is tentative evidence of an increase in the numbers who hardly ever read for pleasure.

Recent years have seen an increased digitalisation of day-to-day lives among adolescents and their families (see, for example, Hartas, 2020). At the age of nine, mobile phone ownership increased significantly between Cohort '98 and Cohort '08. However, mobile phone ownership at the age of 13 was already near-universal for Cohort '98.¹⁴ Analyses looked at three types of screen time: watching television (even on another device); video/computer gaming; and other screen time. At the age of nine, time-use diary information showed little overall change in total screen time but a shift away from TV towards other types of screen time (Smyth, 2022). The absence of a time-use diary for Cohort '08 at 13 means that analyses cannot examine changes in total screen time. However, there was a substantial shift over time in type of screen time, with a sizeable reduction in TV watching and gaming and an increase in other screen-based activities.

TV watching declined across all social groups, but with less of a decline for professional and graduate families, though from initially lower levels. A similar pattern was found when the cohort members were nine years old, a pattern that may suggest the retention of some communal TV viewing among more advantaged groups. Gaming was much more common among boys than girls, although there was a very slight narrowing of this gap over time. There was a strong social gradient in time on gaming and this remained stable over time. Other screen-based activities show a reversal of the earlier gender gap, towards higher levels among girls in Cohort '08 while the social gradient in time spent narrowed only slightly over time. As at nine years of age, there was a trade-off between screen time and involvement in other activities. Both TV and other screen time were associated with lower levels of engagement in organised sports and hard exercise. Both TV and other screen time

¹³ Some caution is needed because of the explicit inclusion of art and crafts in the wording for Cohort '08.

¹⁴ However, the data do not distinguish between smartphones and other phones.

were associated with lower levels of cultural engagement and involvement in organised groups.

5.4 EDUCATIONAL EXPERIENCES

Cohort '08 members had experienced the full implementation of reform at junior cycle level but had also had massive disruption to their learning during the period of pandemic-related school closures. Furthermore, most of the group made the transition to second-level education over the first period of school closures, with less preparation for the transition than their older counterparts would have had. The study findings show that changes in the content of subjects and approaches to teaching and learning have been largely positive for subject engagement among young people, despite pandemic disruption. In particular, the new Science curriculum's emphasis on active enquiry-based learning and fostering enjoyment of the subject appears to have contributed to increasing interest in the subject.¹⁵ Levels of interest have increased too in English and Maths, though not to the same extent as for Science, though there appears to be some polarisation, with lack of interest increasing slightly for some groups. Only in Irish is there evidence of declining interest, with the proportion finding the subject 'not interesting' increasing over time.

In spite of increased interest in three of the subjects considered, marked gender differences remain, with girls more positive about the language-based subjects of English and Irish and boys more positive about Maths and Science. There is evidence of a widening gender gap over time in the latter pattern, which is consistent with the widening of the gender gap in Maths found among these young people at the age of nine. There are also social gradients in attitudes to Maths and Science that are stable over time.

Increased subject engagement does not, however, appear to translate into better school engagement among adolescents. There is, in fact, a decline in the proportion reporting that they like school very much, a pattern that is accounted for by female trends. Preliminary analyses suggest that some of this change is related to increasing emotional difficulties, particularly among girls, post-pandemic. Whether this is due to the disruption of the pandemic and/or to other social changes such as increased screen time and social media engagement merits further research.

5.5 IMPLICATIONS FOR POLICY

The study findings point to an improvement over time in many aspects of young people's lives, particularly more democratic family relations, improved engagement in sport and physical exercise, and greater interest in some core subjects in the wake of junior cycle reform. On the other hand, young people are

¹⁵ Changes in the approach to teaching Science at primary level may also have played a part.

reported to have poorer relations with peers and smaller friendship groups and there is tentative evidence of a growing number who rarely, if ever, read for pleasure.

While there has been a general improvement in many domains, the study findings highlight persistent differences in the lives led by different groups of adolescents, with further research needed to unpack the risk and protective factors underlying these patterns. There has been a good deal of policy development around the new model of Supporting Parents (DCEDIY, 2022). The study findings suggest the need to target supports towards families of children and teenagers with a disability, given the higher levels of parent-child conflict evident in these contexts. Financial strain continues to be a source of friction between parents and teenagers, reinforcing the need to target adequate levels of income support towards families with children to reduce conflict and improve wellbeing.

As at the age of nine (and even earlier, see Smyth, 2016), young people from more disadvantaged backgrounds are less likely to take part in some kinds of out-ofschool activities, including sport and other forms of hard exercise, cultural engagement and, especially among boys, reading for pleasure. This pattern is likely to contribute to an ongoing social gap in cognitive and physical outcomes among adolescents. The findings therefore point to the need for subsidised activities in communities and supports for schools to provide access to a range of extracurricular options. The low levels of involvement in sport and physical exercise among young people with a disability suggest the importance of inclusive practice in out-of-school provision. There is need too to address gender differences in young people's out-of-school activities; there has been increasing policy emphasis on the encouragement of sports involvement among women and girls (Woods et al., 2022) but this could be usefully placed in the context of an emphasis on tackling gender stereotyping across the full range of out-of-school activities. Given that these gendered patterns emerge early (Smyth, 2016), in- and out-of-school settings should seek to provide all young people with access to a range of activities from their early years onwards. As at the age of nine, young people with a disability have much poorer outcomes across the main domains studied here. Further research is planned on the increase over the decade in the proportion with a disability and the implications for their experiences and outcomes.

The findings point to increased interest in English, Maths and, especially, Science (but not Irish) in the wake of junior cycle reform. However, greater interest in these subjects has not translated into improved attitudes to school overall. Indeed, there is a decline in the proportion of girls who like school very much, which preliminary analysis suggests is related to increased emotional difficulties. The social and gender gradient in interest in Maths and Science, and the widening gender gap in STEM interest, is concerning and suggests the need for more inclusive teaching and learning approaches, and, at senior cycle, more inclusive provision, in these subjects.

The study was not designed to look at the effects of the pandemic on adolescent outcomes and it is not possible to determine whether any such effects were temporary or longer lasting. Nonetheless, the findings point to poorer peer relations and more emotional difficulties among this cohort, especially among girls. There is tentative evidence that this gendered pattern of emotional difficulties is linked to greater relative conflict with parents and less positive views of school among girls. Second-level school principals report much poorer wellbeing and school attendance, and, especially in schools serving more disadvantaged communities, poorer school engagement among their students in the post-pandemic period (Smyth, 2023). Further research is merited to identify the school-and family-level factors that have contributed to post-pandemic recovery in order to inform policy development.

REFERENCES

- Altintas, E. and O. Sullivan (2017). 'Trends in fathers' contribution to housework and childcare under different welfare policy regimes', *Social Politics*, Vol. 24, No. 1, pp.81–108.
- Brown, B.B. and J. Larson (2009). 'Peer relationships in adolescence', Handbook of Adolescent Psychology, Vol. 2, Pt. I, pp.74–104.

Chambers, D. and P. Gracia (2021). A Sociology of Family Life. Bristol: Polity Press.

- Conger, K.J., M.A. Rueter and R.D. Conger (2000). 'The role of economic pressure in the lives of parents and their adolescents: the family stress model', *Negotiating Adolescence in Times of Social Change*, L.J. Crockett and R.K. Silbereisen (Eds.), Cambridge University Press, pp.201–223.
- Crockett, L.J. and R.K. Silbereisen (Eds.) (2000). *Negotiating Adolescence in Times of Social Change*, Cambridge University Press.
- De Looze, M.E., A.P. Cosma, W.A. Vollebergh, E.L. Duinhof, S.A. de Roos, S.A.F.M. van Dorsselaer, and G.W.J.M. Stevens (2020). 'Trends over time in adolescent emotional wellbeing in the Netherlands, 2005–2017: links with perceived schoolwork pressure, parent-adolescent communication and bullying victimization', *Journal of Youth and Adolescence*, Vol. 49, No. 10, pp.2124–2135.
- Department of Children, Equality, Disability, Integration and Youth (DCEDIY) (2022). Supporting Parents: A National Model of Parenting Support Services. Dublin: DCEDIY.

Department of Education and Skills (DES) (2015). Framework for Junior Cycle. Dublin: DES.

- Donohue, B., R. Perkins, T. Walsh, B. O'Neill, C. Ó Duibhir, and A. Duggan (2023). *Education in a Dynamic World: the performance of students in Ireland in PISA 2022*. Dublin: Educational Research Centre.
- Dooley, B., C. O'Connor, A. Fitzgerald, and A. O'Reilly (2019). *My World Survey 2: The National Study of Youth Mental Health in Ireland*. Dublin: UCD/Jigsaw.
- Dubas, J.S. and J.R. Gerris (2002). 'Longitudinal changes in the time parents spend in activities with their adolescent children as a function of child age, pubertal status and gender', *Journal of Family Psychology*, Vol. 16, No. 4, p.415.
- Gibbons, R.A., S. Sprong, and Y. Chzhen (2023). 'Growing Up in the Great Recession: the effects of three dimensions of economic well-being on child behavioral difficulties from ages 3 to 17', *Journal of Youth and Adolescence*, Vol. 52, No. 5, pp.1024–1038.
- Gracia, P. (2023). Child and Adolescent Time Use and Well-Being: A Study of Current Debates and Empirical Evidence, OSF Pre-prints.
- GUI Study Team (2021). *Growing Up in Ireland Key Findings: Special COVID-19 Survey*. Dublin: ESRI/TCD/DCEDIY.
- Hagell, A., S.C. Peck, N. Zarrett, J.I. Giménez-Nadal, and J. Symonds (2012). 'Trends in adolescent time use in the United Kingdom', *Changing Adolescence*, pp.47–74.
- Hartas, D. (2020). Young people's play, wellbeing and learning: Psycho-social and virtual geographies in digital play, Springer Nature.

- Högberg, B., M. Strandh, and C. Hagquist (2020). 'Gender and secular trends in adolescent mental health over 24 years – the role of school-related stress', *Social Science & Medicine*, Vol. 250, 112890.
- Hu, Y., and Y. Qian (2021). 'COVID-19 and adolescent mental health in the United Kingdom', Journal of Adolescent Health, Vol. 69, No. 1, pp.26–32.
- Jindal-Snape, D., E.F. Hannah, D. Cantali, W. Barlow, and S. MacGillivray (2020). 'Systematic literature review of primary-secondary transitions: International research', *Review of Education*, Vol. 8, No. 2, pp.526–566.
- Jindal-Snape, D., J.E. Symonds, E.F. Hannah, and W. Barlow (March 2021). 'Conceptualising primary-secondary school transitions: A systematic mapping review of worldviews, theories and frameworks', *Frontiers in Education*, Vol. 6, p.52.
- Larson, R.W., M.H. Richards, G. Moneta, G. Holmbeck, and E. Duckett (2013). 'Changes in adolescents' daily interactions with their families from ages 10 to 18: Disengagement and transformation', *Adolescents and Their Families*, Routledge, pp.118–128).
- Laurence, J., H. Russell, and E. Smyth (2023). *Housing adequacy and child outcomes in early and middle childhood*, ESRI Research Series No. 154, Dublin: The Economic and Social Research Institute, https://doi.org/10.26504/rs154.
- Luiten, A., J. Hox, and E. de Leeuw (2020). Survey nonresponse trends and fieldwork effort in the 21st century: Results of an international study across countries and surveys, *Journal of Official Statistics*, Vol. 36, No. 3, pp.469–487.
- McGarr, O., O. McCormack, J. O'Reilly, R. Lynch, J. Power, J. Hennessy, N. O'Meara, A. Neary, K. Leahy, A. Calderón, A. MacPhail, C. Ó Gallchóir, J. McMahon, R. Lenihan, C. Ní Chathasaigh, and M. Goos (2023). *Exploring the introduction of the Framework for Junior Cycle: A longitudinal study, Interim report No. 2 (Version 1)*, Limerick: University of Limerick.
- McGinnity, F., S. Enright, E. Quinn, B. Maître, I. Privalko, M. Darmody, and M. Polakowski (2020). *Monitoring Report on Integration 2020*, Dublin: The Economic and Social Research Institute, https://doi.org/10.26504/bkmnext403.
- Mullan, K. (2019). 'A child's day: trends in time use in the UK from 1975 to 2015', *The British Journal of Sociology*, Vol. 70, No. 3, pp.997–1024.
- Murray, A., D. Watson, A. Nolan, B. Duggan, E. McNamara, D. O'Mahony, E. Smyth, and A. Quail (2023). *Design, Instrumentation and Procedures for Cohort '08 of Growing Up in Ireland at 13 Years Old (Wave 6)*, Dublin: ESRI/TCD/DCEDIY.
- National Council for Curriculum and Assessment (NCCA) (2021). *Wellbeing Guidelines,* Dublin: NCCA.
- Newlove-Delgado, T., S. McManus, K. Sadler, S. Thandi, T. Vizard, C. Cartwright and T. Ford (2021). 'Child mental health in England before and during the COVID-19 lockdown', *The Lancet Psychiatry*, Vol. 8, No. 5, pp.353–354.
- Orben, A. and A.K. Przybylski (2019). 'Screens, teens, and psychological well-being: Evidence from three time-use-diary studies', *Psychological Science*, Vol. 30, No. 5, pp.682–696.

- Patalay, P. and S.H. Gage (2019). 'Changes in millennial adolescent mental health and health-related behaviours over 10 years: a population cohort comparison study', *International Journal of Epidemiology*, Vol. 48, No. 5, pp.1650–1664.
- Reinhard, E., R. Layte, C. McCrory, L. Panico and M. Avendano (2018). 'The Great Recession and the Health of Young Children: A fixed-effects analysis in Ireland', American Journal of Epidemiology, Vol. 187, No. 7, pp.1438–1448.
- Ryan, R.M., A. Kalil, C. Hines and K. Ziol-Guest (2020). 'Trends in parental values in a period of US labor market change', *Journal of Marriage and Family*, Vol. 82, No. 5, pp.1495–1514.
- Sayer, L.C., S.M. Bianchi and J.P. Robinson (2004). 'Are parents investing less in children? Trends in mothers' and fathers' time with children', *American Journal of Sociology*, Vol. 110, No. 1, pp.1–43.
- Schoon, I. (2006). *Risk and resilience: Adaptations in changing times,* Cambridge: Cambridge University Press.
- Smyth, E. (2016). Arts and Cultural Participation Among Children and Young People: Insights from the Growing Up in Ireland Study, Dublin: The Economic and Social Research Institute.
- Smyth, E. (2022). *The Changing Social Worlds of 9-Year-Olds,* ESRI Research Series No. 151, Dublin: The Economic and Social Research Institute, https://doi.org/10.26504/rs151.
- Smyth, E. (2023). *Child Wellbeing in the Wake of the Pandemic: Implications for the School Completion Programme*, Presentation to the SCP National Conference.
- Smyth, E., S. McCoy and M. Darmody (2004). *Moving Up: The experiences of first-year students in post-primary education*, Dublin: Liffey Press.
- Smyth, E. and A. Murray (2022). 'The Effect of Pandemic-Related Economic Disruption on Young Adolescents in Ireland', *Children*, Vol. 9, No. 7, p.1037.
- Smyth, E. and H. Russell (2021). *Fathers and Children from Infancy to Middle Childhood,* ESRI Research Series 130, Dublin: The Economic and Social Research Institute, https://doi.org/10.26504/RS130.
- Sprong, S., R.A. Gibbons and Y. Chzhen (2023). 'Divergent trajectories: three dimensions of child poverty during the Great Recession in Ireland', *Longitudinal and Life Course Studies*, Vol. 14, No. 1, pp.128–137.
- Thornton, M., J. Williams, C. McCrory, A. Murray and A. Quail (2016). *Design, Instrumentation and Procedures for the Child Cohort at Wave Two (13 Years),* Dublin: ESRI/TCD/DCYA.
- Twenge, J.M. and G.N. Martin (2020). 'Gender differences in associations between digital media use and psychological well-being: Evidence from three large datasets', *Journal of Adolescence*, Vol. 79, No. 1, pp.91–102.
- Twenge, J.M., G.N. Martin and B.H. Spitzberg (2019). 'Trends in US Adolescents' Media Use, 1976–2016: The rise of digital media, the decline of TV, and the (near) demise of print', *Psychology of Popular Media Culture*, Vol. 8, No. 4, p.329.

- Woods C.B., K.W. Ng, U. Britton, J. McClelland, B. O'Keeffe, A. Sheikhi, P. McFlynn, M.H. Murphy, H. Goss, S. Behan, C. Philpott, D. Lester, M. Adamakis, J. Costa, T. Coppinger, S. Connolly, S. Belton and W. O'Brien (2022). *The Children's Sport Participation and Physical Activity Study 2022 (CSPPA 2022)*, Limerick: University of Limerick and Belfast: Sport Northern Ireland.
- Zimmer-Gembeck, M.J. and W.A. Collins (2006). 'Autonomy Development During Adolescence', *Blackwell Handbook of Adolescence*, pp.174–204.

Whitaker Square, Sir John Rogerson's Quay, Dublin 2 Telephone **+353 1 863 2000** Email **admin@esri.ie** Web **www.esri.ie** Twitter **@ESRIDublin**

